2023, Vol. 149, Nos. 7-8, 418-446 https://doi.org/10.1037/bul0000401

The Development of Subjective Well-Being Across the Life Span: A Meta-Analytic Review of Longitudinal Studies

Susanne Buecker^{1, 2}, Maike Luhmann¹, Peter Haehner¹, Janina Larissa Bühler³,

Laura C. Dapp⁴, Eva C. Luciano⁵, and Ulrich Orth⁴

¹ Faculty of Psychology, Ruhr University Bochum ² Institute of Psychology, German Sport University Cologne ³ Institute of Psychology, Johannes Gutenberg University Mainz

⁴ Institute of Psychology, University of Bern
 ⁵ Faculty of Psychology, University of Basel

How does subjective well-being (SWB) develop across the life span? Theories and previous empirical research suggest heterogeneous conclusions regarding this question. Therefore, in this meta-analysis, we synthesized the available longitudinal data on mean-level change in three SWB components: life satisfaction, positive affect, and negative affect. The analyses were based on 443 unique samples with a total of 460,902 participants. Our results showed that life satisfaction decreased from age 9 to 16 (d = -0.56), increased slightly until age 70 (d = 0.16), and then decreased again until age 96 (i.e., the oldest age for which data on life satisfaction were available; d = -0.24). Positive affect declined from age 9 for almost the entire time until age 94 (d = -1.71). Negative affect showed small ups and downs between ages 9 and 22. After age 22, negative affect declined until age 60 (d = -0.92), after which it increased again until age 87 (d = 0.58). Average changes in positive and negative affect were stronger than in life satisfaction. The moderator analyses suggested that the pattern of mean-level changes held across gender, country, ethnicity, sample type, the measure of SWB, time frame of SWB measure, and birth cohort. In sum, we found a favorable developmental trajectory of SWB over large parts of life for life satisfaction and negative affect and decreases from childhood until late adulthood for positive affect. In late adulthood, SWB tended to worsen rather than improve. Consequently, interventions aimed at maintaining or enhancing SWB in older adults might be useful.

Public Significance Statement

This meta-analysis of longitudinal studies suggests that the three components of subjective well-being (i.e., life satisfaction, positive affect, and negative affect) follow distinct developmental patterns across the life span. Whereas life satisfaction and negative affect showed favorable developmental trajectories over large parts of life (i.e., from adolescence until about 70 years), positive affect decreased from childhood until late adulthood.

Keywords: subjective well-being, life satisfaction, affect, life span development, meta-analysis

Supplemental materials: https://doi.org/10.1037/bul0000401.supp

At what age are people happiest? This seemingly simple question has been investigated extensively by social scientists over the last decades, but a conclusive answer has yet to be found. To date, hundreds of cross-sectional and longitudinal studies on the development of happiness have been published, with samples from many different countries and generations. In most of these studies, happiness was conceptualized in terms of *subjective well-being* (SWB), which comprises life satisfaction, positive affect, and

Susanne Buecker https://orcid.org/0000-0003-3443-5400 Maike Luhmann https://orcid.org/0000-0001-6211-9304 Peter Haehner https://orcid.org/0000-0002-3896-6172 Janina Larissa Bühler https://orcid.org/0000-0003-3684-9682 Laura C. Dapp https://orcid.org/0000-0003-3985-6976 Ulrich Orth https://orcid.org/0000-0002-4795-515X

A networking and travel stipend of the European Association of Personality Psychology to Susanne Buecker was used to prepare this work. Moreover, this research was supported by the Swiss National Science Foundation Grant P2BSP1_188102 to Janina Larissa Bühler. The authors thank Elena Dick, Anina Flück, Nicole Schaub, and Jeanine Wyss for their help in the early phase of the data collection. The authors have no conflicts of interest to disclose.

The authors followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses guidelines. The coding instructions, data sets, and R code files used for the analyses are available at OSF (https://osf.io/wn7vm/; Buecker et al., 2023). This research was not preregistered.

Correspondence concerning this article should be addressed to Susanne Buecker, Institute of Psychology, German Sport University Cologne, Am Sportpark Müngersdorf 6, 50933 Cologne, Germany. Email: s.buecker@dshs-koeln.de negative affect (Diener, 1984). Despite the extensive studies, the typical shape of the developmental trajectory of SWB across the life span is still under debate. Some studies reported that SWB follows a linear trajectory across the life span (i.e., showing increases or decreases with age; e.g., Pinquart, 2001). Others found that SWB declines particularly in late adulthood during the last years before death (Gerstorf et al., 2008). Finally, several (mostly cross-sectional) studies suggested a U-shaped trajectory across the life span (i.e., high levels of SWB in young adulthood, a nadir in middle adulthood, and high levels of SWB in late adulthood; e.g., Blanchflower & Graham, 2021a; Blanchflower & Oswald, 2008).

The strong scientific interest in the developmental trajectory of SWB across the life span has numerous reasons. When asking people of different ages to list the characteristics they value most in life, they are likely to include aspects of SWB (e.g., "being happy"; King & Napa, 1998; Luhmann et al., 2014). Furthermore, SWB has been found to predict health and longevity (Diener & Chan, 2011; Howell et al., 2007). Due to the desirable consequences of SWB for individuals, many governments view SWB as an important input to policy decisions and include national accounts of SWB as an indicator of a nation's wealth (Diener et al., 2015). It is important to understand whether and how SWB changes as a (possibly nonlinear) function of age when interpreting findings of the individual and societal consequences of SWB. In this way, it is possible to identify age groups that are particularly at risk for loss in SWB and associated negative consequences and to identify countries whose demographic changes contribute to an increased risk for reduced SWB in the population.

Yet, findings on the developmental trajectory of SWB are heterogeneous. In such a situation, comprehensive meta-analyses are ideal for advancing the state of knowledge about a phenomenon. Therefore, in the present meta-analysis, we synthesized the available longitudinal data on mean-level changes in SWB across the life span. Besides estimating the average mean-level change in SWB, we tested potential moderators of these changes.

Definition and Measurement of SWB

According to the widely used definition by Diener et al. (1999), SWB comprises cognitive and affective components. Cognitive well-being includes general life satisfaction (i.e., the cognitive evaluation of one's life overall) and domain satisfaction (i.e., the cognitive evaluation of specific life domains such as job, family, or health). Yet, most studies on SWB only investigate general life satisfaction, but not domain satisfaction. Affective well-being includes positive affect (i.e., the frequency and intensity of positive emotions, such as feeling interested, attentive, and joyful as well as more general positive mood states) and negative affect (i.e., the frequency and intensity of negative emotions, such as feeling frightened, angry, and sad as well as more general negative mood states).

The SWB components are correlated but conceptually and empirically distinct. For example, major life events tend to have more long-lasting effects on cognitive well-being than on affective well-being (Luhmann, Hofmann, et al., 2012). In addition, although both cognitive and affective components of SWB are correlated with personality traits, such as extraversion and neuroticism, the correlations tend to be stronger for affective well-being than for cognitive well-being (e.g., Soto, 2015). Cognitive well-being, in contrast, tends to be more strongly associated with life circumstances such as income, work status, or marital status than affective well-being (Luhmann et al., 2014; Schimmack et al., 2008; Tov, 2018). Within affective well-being, positive and negative affect are treated either as separate constructs or as opposite poles of a single dimension. Overall, research on the structure of SWB (e.g., Lucas et al., 1996; Luhmann, Hofmann, et al., 2012; Tov, 2018) indicates that the different components should be measured and examined separately, as we did in this meta-analysis.

All components of SWB can be conceptualized as being relatively stable across time and consistent across different contexts (e.g., feeling satisfied with one's work and family life). This perspective refers to habitual or trait-like SWB (Brose et al., 2013). Nevertheless, all components of SWB can also be conceptualized as fluctuating from day to day or from moment to moment, which is typically described as state-like SWB (Brose et al., 2013). Depending on which conceptualization of SWB is used, studies differ in the time frame that is used to measure SWB. Most studies measuring life satisfaction use a rather broad time frame asking about how satisfied one is in general. In contrast, positive and negative affect are typically measured using specific and typically short time frames (e.g., during the past few weeks e.g., Burr et al., 2011) or during the past few months (e.g., Müller et al., 2014). Therefore, in the present meta-analysis, we included these different time frames as a potential moderator of mean-level change in SWB across the life span (see below for more information).

Beyond life satisfaction, positive affect, and negative affect, some studies examined what was referred to as "happiness". Happiness is less consistently defined and measured than SWB. When people say that they lead a "happy life", they typically refer to a broad evaluation of one's life as a whole and thus report on the cognitive part of well-being. Still, there are other meanings of happiness that refer to a particular affective experience that happens over a short time, such as being happy while celebrating a party, chatting with friends, or doing other enjoyable activities. These short-term experiences may not reflect the broader evaluation of one's life. Given the many different terms and conceptualizations used by wellbeing researchers, we focused on those constructs that can clearly be assigned to SWB according to Diener (1984): life satisfaction, positive affect, and negative affect. In this way, we ensured that the results of the meta-analysis allow for conclusions about clearly defined constructs.

Previous Research on the Development of SWB Across the Life Span

Among the heterogeneous findings on the development of SWB across the life span, the findings suggesting a U-shaped trajectory have elicited much debate in the literature (e.g., Blanchflower & Graham, 2021a; Galambos et al., 2020, 2021). Although many studies support the idea of a U-shaped trajectory of SWB across the life span (e.g., Blanchflower & Graham, 2021a, 2021b; Blanchflower & Oswald, 2008), there are several caveats regarding these studies.

First, the U-shaped trajectory cannot be found for all three components of SWB. Studies that suggested a U-shaped trajectory usually focused on life satisfaction (Kolosnitsyna et al., 2017; Otterbach et al., 2018; Piper, 2015) or positive affect (e.g., Blanchflower & Graham, 2021a; Blanchflower & Oswald, 2008; but see Hudson et al., 2016). In contrast, for negative affect, a

linear decrease appears to be the more common finding (Galambos et al., 2006). To get a comprehensive picture of how SWB develops across the life span, it is necessary to examine all three components of SWB.

Second, many studies that found a U-shaped trajectory used cross-sectional data. Unfortunately, cross-sectional studies confound age and cohort effects (Baltes et al., 1979; Costa & McCrae, 1982). Thus, it is unclear whether the U-shaped relation between age and SWB can be explained by developmental processes related to age or rather by specific life experiences that distinguish some generations from others. Age and cohort effects can be disentangled in studies that track multiple cohorts across time, as was done in this meta-analysis. Longitudinal data are generally more appropriate to study developmental trajectories (Kraemer et al., 2000; Kratz & Brüderl, 2021). For SWB, it seems that the average trajectories found in longitudinal studies often diverge from those observed in crosssectional studies. For example, a longitudinal study with healthy male veterans showed an increase in life satisfaction from about 40 to 65 years, followed by a decrease (Mroczek & Spiro, 2005), which is inconsistent with the U-shaped trajectory reported in many crosssectional studies.

Third, a further complication is that the findings might also depend on the data-analytic approach employed in a study (Kratz & Brüderl, 2021). Using the same large-scale longitudinal data set (i.e., the German Socio-Economic Panel; Goebel et al., 2018) but different data-analytic methods, researchers have documented U-shaped, S-shaped, linear upward, and linear downward developmental trajectories for life satisfaction across the life span (Baetschmann, 2014; Baird et al., 2010; Bauer et al., 2017; Cheng et al., 2017; Wunder et al., 2013). Another methodological challenge is the different treatment of control variables between different studies examining how age affects SWB. When integrating relevant control variables (i.e., those influencing the population's age composition and SWB), no U-shaped trajectory was found (Bartram, 2021). Moreover, recent research emphasized turning points of SWB around midlife (mid-40s) and in late adulthood (90s) when looking at between-person results (Biermann et al., 2022). Yet, in studies focusing on within-person changes, SWB appeared mostly stable between ages 16 and 23 and then approached a local maximum at age 75 (if the same person is followed over time; Biermann et al., 2022).

Fourth, most previous studies used North American samples. Still, some studies suggest that the typical trajectory of life satisfaction may be different for samples from other parts of the world (Baird et al., 2010). Thus, it is important to examine the development of SWB in different countries and cultural contexts, as we did in this meta-analysis.

Yet, despite these heterogeneous results, two findings have been replicated in multiple studies: Life satisfaction tends to increase from midlife to late adulthood (Lachman et al., 2008; Mroczek & Spiro, 2005; Stone et al., 2010), and life satisfaction and positive affect tend to decline in very late adulthood (Baird et al., 2010; Berg et al., 2009; Gerstorf et al., 2008; Mroczek & Spiro, 2005).

Theoretical Perspectives on the Development of SWB Across the Life Span

To date, no theory is available that fully explains how SWB changes across the life span. Nevertheless, several theories from the fields of positive psychology, personality psychology, and developmental psychology allow deriving hypotheses about the trajectories of SWB across the life span. In the following, we summarize the core principles of theoretical perspectives that are relevant for understanding the development of SWB and outline similarities and differences among these theories. Not all of the reviewed theories make direct statements about SWB; in some cases, they refer to SWB-related constructs, such as personality traits or emotion regulation, which can provide insights into the developmental trajectory of SWB. Moreover, we describe life circumstances or major life events that are linked to SWB and occur more often in specific life stages than in others (i.e., differences regarding the prevalence). We briefly review that some life circumstances or major life events might be especially important for SWB in specific life stages (i.e., differences regarding the relevance).

Set Point Theory

Early research on SWB has theorized that SWB is a very stable characteristic of individuals: Some people are born satisfied, and others are not (e.g., Brickman & Campbell, 1971). Consistent with this idea are findings showing that SWB is, to some extent, heritable. For example, two meta-analyses of twin and family studies have estimated an average heritability of 36% (Bartels, 2015) and 40% (Nes & Røysamb, 2015). Set point theory—which is sometimes also referred to as adaptation-level theory (Diener et al., 2006)-suggests that major life events (e.g., marriage, divorce, becoming unemployed) may lead to temporary increases or decreases in SWB but that psychological mechanisms regulate SWB such that it eventually returns to its set point (Lykken & Tellegen, 1996). According to the set point theory, SWB should be highly stable across the life span. In recent years, however, the idea of an immutable set point of SWB has been challenged by many empirical studies (Diener et al., 2006; for an overview, see Hudson et al., 2019).

Neo-Socioanalytic Theory

Neo-socioanalytic theory suggests that adults typically develop in the direction of mature personality traits, including a decrease in neuroticism, especially in young adulthood (Roberts & Robins, 2021; Roberts & Wood, 2006). This so-called maturity principle has been supported in numerous studies (Bleidorn et al., 2009; Caspi et al., 2005; Roberts et al., 2006; Specht et al., 2011). It has been argued that the adoption of age-related social roles (e.g., the role of an employee when entering the job market or the role of a mother when getting one's first child) leads to the specific behavior that is required to better fulfill these social roles (Bleidorn, 2015; Roberts & Wood, 2006). These behavioral changes are then reflected in changes in broader personality traits. Yet, recent studies on the development of personality traits and socioemotional phenomena surrounding major life events have challenged this view because they did not find evidence for all hypothesized changes (Denissen et al., 2019; Hang et al., 2023).

As neuroticism is closely linked to negative affect (DeNeve & Cooper, 1998), neo-socioanalytic theory suggests that negative affect decreases in adulthood, especially in young adulthood. For late adulthood, however, research has found that social vitality (i.e., a facet of extraversion) decreases. Since social vitality is positively related to positive affect and to life satisfaction but negatively to

negative affect (DeNeve & Cooper, 1998), this finding suggests based on neo-socioanalytic theory—that positive affect and life satisfaction might decrease in late adulthood, while negative affect might increase.

Disruption Hypothesis

Whereas the maturity principle has received relatively strong empirical support for adult personality development, in youth personality development, another finding is striking: Certain desirable personality traits, including extraversion and emotional stability (i.e., low neuroticism), show a temporary dip from childhood to adolescence (Brandes et al., 2021; Soto, 2016). This trend has been called the disruption hypothesis (Soto & Tackett, 2015). Girls especially appear to become prone to negative affect during adolescence, whereas for boys, negative affect was rather stable (Soto et al., 2011; Van den Akker et al., 2014). Moreover, levels of sociability are lower during childhood and adolescence than during adulthood (Denissen et al., 2013; Van den Akker et al., 2014). As noted above, SWB is correlated with neuroticism and extraversion. Thus, the disruption hypothesis suggests that life satisfaction and positive affect decrease from childhood to adolescence, while at the same time, negative affect increases. Moreover, there might be gender differences-especially in meanlevel change of negative affect during childhood and adolescence.

Socioemotional Selectivity Theory

Whereas research on the disruption hypothesis allows deriving hypotheses about the development of SWB in youth, socioemotional selectivity theory might be relevant for understanding changes in SWB in later life. This theory suggests that with increasing age—once people realize that their remaining lifetime is limited—people select themselves into situations that maximize the experience of positive emotions and minimize the experience of negative emotions (Carstensen et al., 1999). Moreover, when such a selection of situations is not possible, older adults show enhanced emotion regulation by selecting and optimizing emotion regulation processes to compensate for losses in internal and external resources (Urry & Gross, 2010). Consequently, socioemotional selectivity theory suggests that affective well-being (more than cognitive well-being) changes in older adulthood in a way that leads to increases in positive affect and decreases in negative affect.

Theory of Dynamic Integration

Both the optimization of happiness and positive affect as well as the ability to tolerate and differentiate tension and negative affect are considered crucial for high well-being. The balance between optimization and differentiation is called *dynamic integration* (Labouvie-Vief, 2003). The capacity to dynamically integrate increases from adolescence to middle adulthood, possibly because individuals acquire more conscious insight into their emotions and learn to differentiate blended distinct emotions (e.g., emotions involving both positive and negative affective states such as sadness and joy; Labouvie-Vief, 2003). This theoretical perspective is supported by empirical findings on growth in affective complexity (i.e., a better understanding and clearer description of one's own and other's emotions) through middle adulthood (Labouvie-Vief et al., 2007). Still, dynamic integration decreases later in life, possibly due to cognitive decline and poor emotion regulation strategies (Grühn et al., 2013; Labouvie-Vief et al., 2007). Older adults may face difficulties in the integration and toleration of negative affect. While dynamic integration theory does not provide a clear prediction for how positive affect, negative affect, and life satisfaction develop across the life span, the theory does emphasize that the nature of emotion regulation and the integration of emotions and cognitions change across the life span. The theory of dynamic integration discusses these changes as mechanisms for changes in well-being.

Model of Strength and Vulnerability Integration

The Model of Strength and Vulnerability Integration (SAVI; Charles, 2010) describes changes in the processes of emotion regulation in later adulthood, which are used to explain changes in affective well-being from middle to later adulthood. The model distinguishes between age-related strengths and vulnerabilities. On the one hand, aging is related to increased strengths in the frequency and successful use of adaptive emotion regulation strategies such as appraisals. These strengths allow people to circumvent negative emotions and enhance positive emotions. The increase in such emotion regulation processes may explain why some studies observed high levels of affective well-being in late adulthood. On the other hand, aging is also related to vulnerabilities such as reduced physiological flexibility. The SAVI model postulates that if older adults can employ the strengths of aging, age-related increases in affective well-being will emerge. Though, if age-related vulnerabilities predominate, affective well-being is expected to decline. It is assumed that some situations happen in late adulthood where there is little opportunity to appraise the situation and eliminate negative feelings or to direct attention elsewhere (i.e., the strengths of age cannot be exploited), resulting in decreases in well-being. Such situations include threats and loss of social belonging, exposure to uncontrollable chronic stressors, and neurological dysregulation (Charles, 2010).

Terminal Decline Hypothesis

SWB shows steep end-of-life deterioration—a phenomenon called *terminal decline* (Gerstorf et al., 2008; Gerstorf & Ram, 2015). The terminal decline hypothesis postulates that a preterminal phase of gradual decline is followed by a terminal phase of much more pronounced decline in the last years before death, for example, due to changes in social orientations and social losses (Gerstorf et al., 2008, 2016). For instance, using 22-wave longitudinal data, Gerstorf et al. (2008) showed that the decline in life satisfaction steepened about 4 years prior to death. Moreover, Vogel et al. (2013) found decreases in positive affect and increases in negative affect accelerating shortly before death. Consequently, the terminal decline hypothesis suggests a substantial decline in life satisfaction and positive affect and a substantial increase in negative affect in late adulthood.

Changes in Life Circumstances as Triggers of Changes in SWB

People's objective life circumstances (e.g., health, relationship status, work status) change in normative ways across the life span (e.g., Alwin & Wray, 2005; Dolan & Sale, 2019; Hutteman et al.,

2014; Wrzus, Hänel, et al., 2013). For example, most people start their first romantic relationship during late adolescence or young adulthood (Boisvert & Poulin, 2016) and enter the job market during young adulthood (Pusch et al., 2019). There is strong evidence that these life circumstances are related to SWB (e.g., Braithwaite et al., 2010; Dush & Amato, 2005; Lucas et al., 2004; Lukkala et al., 2016). Even still, certain life circumstances are more prevalent (i.e., occur more frequently) or more relevant (i.e., are more closely related to SWB) during some life stages than during others. Consequently, changes in SWB across the life span could at least partially be explained by changes in the prevalence and the relevance of such life circumstances (e.g., declining health in late adulthood; Kunzmann et al., 2000). For instance, young adulthood is characterized by a large number of major life events, most of which are positively perceived (Lüdtke et al., 2011), resulting in higher SWB during this life stage. In contrast, late adulthood is characterized by more death- and loss-related life events (e.g., death of peers or partner; Buecker et al., 2021; Luhmann, Hofmann, et al., 2012), which may result in lower SWB during this life stage.

Moreover, the relative importance of certain life circumstances for SWB varies systematically across the life span (Cheung & Lucas, 2015; George et al., 1985). According to the motivational theory of life span development (Heckhausen et al., 2010), different life stages are associated with unique demands and expectations, also called developmental tasks. People are motivated to invest their resources to fulfill the key developmental tasks of their life stage. For example, when people perceive certain life circumstances as important and desirable during a certain life stage (e.g., starting a family in young adulthood) and can achieve these life circumstances, this might enhance their SWB in this life stage. Yet, the same life circumstances might not have a strong beneficial effect during other life stages. Moreover, when people cannot reach their desired life circumstances in a certain life stage, the absence of certain events may harm SWB (Luhmann et al., 2021). In summary, changes in SWB across the life span might be partially accounted for by changes in people's actual and desired life circumstances.

Change of Emotional Intensity Across the Life Span

Research suggests that the intensity of general affect (for both positive and negative affect) decreases across the life span (for a review, see Bailen et al., 2019; Larsen & Diener, 1987). For example, compared to an adult sample, adolescents reported more intense positive and negative emotions (Larson et al., 1980). Possible explanations for this decrease in emotional intensity are related to biological changes that occur with age and to the finding that older people have already been exposed to much more emotional incidents and, consequently, have habituated in their emotional responses or have improved their emotion regulation strategies. Indeed, self-report studies show that older adults view themselves as more in control of their emotions (e.g., Lawton, 2001) and show lower physiological responses to mild stressors (e.g., Wrzus, Müller, et al., 2013) than younger adults. Moreover, as people age, they may have learned from their experiences how to cope with difficult situations or major life events and may thus be better able to put their emotions into perspective. The decreasing trend in affect intensity across the life span suggests that both positive and negative affect decrease as a function of age.

Potential Moderators of the Development of SWB

In this meta-analysis, we expected that there is significant heterogeneity in the findings on the development of SWB across the life span. For example, the developmental pattern in SWB might differ across cohorts because each cohort has been raised in a specific sociohistorical context, which might impact the SWB of the cohort members (Bühler & Nikitin, 2020; Drewelies et al., 2019; Gerstorf et al., 2020). Still, thus far, little is known about cohort effects in the development of SWB. Therefore, we tested for cohort effects using the sample's mean year of birth as a proxy for the cohort.¹

Also, it is possible that other sample characteristics account for the heterogeneity of findings. Therefore, we tested whether sample type, ethnicity, country of origin, and gender composition explain differences in the development of SWB across the life span. For example, studies using a nationally representative sample are more generalizable than studies using convenience samples. If there are no significant differences in the effects between these sample types, then this supports the overall robustness of the meta-analytic findings. Moreover, as described above, different life stages are associated with unique demands and expectations, and their fulfillment is related to SWB (Heckhausen et al., 2010). These demands and expectations could vary across ethnic groups and countries (Bleidorn et al., 2013). Nevertheless, in previous primary studies, cultural and ethnic differences in SWB development have oftentimes been neglected. Therefore, we tested whether ethnicity and country of origin moderated the effect sizes. Finally, previous research found gender differences in the development of SWB, especially in childhood and adolescence (González-Carrasco et al., 2017; Steinmayr et al., 2019). Such gender differences might be due to differences in SWB-related biological changes or changes in social roles occurring in these age groups. Therefore, we tested whether the percentage of females in the sample moderated the findings.

Because the field of SWB research uses a wide variety of measures, we tested whether effect sizes from studies using the most prominent measures of life satisfaction (i.e., Satisfaction With Life Scale; Diener et al., 1985) and affect (i.e., Positive and Negative Affect Scale [PANAS]; Watson et al., 1988) differed from studies using other measures, such as selected mood adjectives from Bradburn's affect scales (Bradburn, 1969). Moreover, previous studies discussed the role of different time frames (e.g., referring to affect during the last week, month, or year) when studying SWB and its development (e.g., Luhmann, Hawkley, et al., 2012). Therefore, we also tested whether the time frame moderated the effect sizes.

Besides the moderators that were included in this meta-analysis, there are other characteristics that would also be theoretically relevant but could not be tested. For example, socioeconomic status is related to SWB (Cheung & Lucas, 2015; Sainz et al., 2021) and might also affect the development of SWB across the life span. Regardless, this information was rarely reported and, if reported, the measures were not comparable across most of the included studies.

¹ Note that the samples included in this meta-analysis were relatively homogeneous regarding age, which is a precondition that the mean year of birth of the sample is an adequate approximation of the birth cohort.

The Present Research

The ongoing debate about the typical, normative trajectory of SWB across the life span (e.g., Blanchflower & Graham, 2021a; Galambos et al., 2020, 2021) has highlighted that previous empirical findings and theoretical perspectives on SWB across the life span are inconsistent. Therefore, the goal of the present research was to draw a precise and comprehensive picture of the normative trajectories of life satisfaction, positive affect, and negative affect from childhood to late adulthood by meta-analyzing the available data from longitudinal studies on SWB. Moreover, we tested several moderators of the development of SWB: Birth cohort, sample type, ethnicity, country of origin, gender, the measure of SWB, and time frame of the SWB measure.

Initially, we had planned to include happiness as a separate outcome in the meta-analysis (in addition to life satisfaction, positive affect, and negative affect). Yet, the coding process showed that the number of studies that included happiness as an outcome was too small. Multiple effect sizes per age group would be needed to gain reliable insights into happiness changes across the life span. Unfortunately, only 10 studies were available (across all age groups) that provided data on happiness. Therefore, this outcome was not reported separately in the article. Although, when a study used different adjectives expressing positive affect (e.g., happy, excited, enthusiastic) and reported an average score across these items, happiness was included as a part of positive affect in the metaanalysis.

Method

The present meta-analysis used anonymized data and therefore was exempt from approval by the Ethics Committee of the first author's institution (German Sport University Cologne), in accordance with national law.

Systematic Literature Search

We searched for relevant studies in the databases APA PsycInfo and MEDLINE; the search covered all entries until November 10, 2022. We used the following search terms: life satisfaction OR satisfaction with life OR positive affect* OR negative affect* OR subjective well-being. The use of the asterisks allowed us to identify variants of the respective keyword (i.e., by using the term affect* we could identify studies using the terms affect and affectivity). The search terms were entered without quotation marks so that not only "subjective well-being" but also "well-being" in general (and thus also "psychological well-being") appeared in the search hit list. The terms were searched in the title, abstract, key concepts, and subject headings. In APA PsycInfo, we restricted the search to empiricalquantitative studies and longitudinal studies by using the limitation options "empirical study," "quantitative study," and "longitudinal study." Moreover, we restricted the search to studies using a human population and published them in English. As MEDLINE does not offer such limitation options, we added these methodological terms to our search string. We included only effect sizes from nondisordered populations. Therefore, this search procedure was performed once by setting an APA PsycInfo limit to "nondisordered populations" and once by setting the limit to "disordered populations". As the APA PsycInfo tags on the clinical status of the sample are sometimes faulty

(e.g., a two-study article with one disordered sample and one nondisordered sample is tagged as "disordered sample"), we screened the titles and abstracts of the subsearch on disordered samples regarding the clinical status of the sample. Studies in this subsearch that used at least one nondisordered sample were assessed in full text for eligibility for this meta-analysis. In contrast, studies that used only clinical samples were excluded in this step. The total number of articles found in these two subsearches corresponds to the total number of studies found without setting any limit regarding the clinical status of the sample. In MEDLINE, no limit regarding the clinical status of the sample could be set. Therefore, studies located via MEDLINE were also screened regarding the clinical status, and studies that used only clinical samples were excluded. In total, $k_{\text{total}} =$ 4,754 studies were found using both search engines. The flowchart summarizing the search and selection procedure (Figure 1) shows the number of included and excluded studies.

In addition to the search in databases, we solicited unpublished studies from the scientific community in several psychological subdisciplines (e.g., developmental psychology, personality psychology, gerontology). More specifically, we used mailing lists from various associations such as the Society for the Study of Emerging Adulthood, the Society for Personality and Social Psychology, the European Association of Personality Psychology, and the German Association of Psychology. In total, we included k = 24 additional studies (20 unpublished reports, four published studies) with 114 unique samples.

We assessed the validity of the systematic literature search by checking whether narrative reviews on the development of SWB across the life span (i.e., Galambos et al., 2020; Hudson et al., 2019) included longitudinal studies that were not found in the systematic literature search. The results of this procedure showed that the relevant studies (or, more precisely, the data that were used in the studies) were already included in the meta-analytic data set, strengthening our confidence regarding the comprehensiveness of the systematic literature search.

Inclusion Criteria

We included studies if they fulfilled the following eight criteria. (a) The study assessed SWB, including measures of life satisfaction, positive affect, or negative affect. Measures limited to less than three specific emotions (e.g., anger, guilt, pride), and measures of stress, depressed mood, and psychological well-being were not included. (b) The study included a longitudinal measure of SWB with a time lag of at least 6 months (for a similar procedure, see Orth, Erol, & Luciano, 2018). (c) The measure of SWB was identical across measurement occasions (i.e., regarding the number of items, response scale, item wording, etc.). (d) The standard deviation (SD) of age in the sample was not larger than 5 years at the first measurement occasion to ensure sufficient homogeneity regarding age (for a similar procedure, see Orth, Erol, & Luciano, 2018). If the heterogeneity regarding age is strong, it is unclear whether the observed mean-level change can be validly related to the average age in the sample. For example, when considering a sample with a broad age range (e.g., 18-60 years), a mean age of 32 years, a SD of the age of 10 years, and that shows a mean-level decrease in a construct of d = -0.20 across a 1-year interval, it is unclear whether the effect size (d = -0.20) could capture normative change at the mean age (32 years) with sufficient precision. Particularly if the age

Content may be shared at no cost, but any requests to reuse this content in part or whole must go through the American Psychological Association. This document is copyrighted by the American Psychological Association or one of its allied publishers.

Figure 1

Adapted PRISMA 2020 Flow Diagram (Page et al., 2021) Visualizing the Search and Coding Procedures



SWB = subjective well-being; k = the number of records (i.e., studies); PRISMA = Preferred Reporting Items for Systematic Reviews and Meta-Analyses. See the online article for the color version of this figure. Note.

range of the sample covers developmental stages with distinct patterns of normative change, the average change in the construct could be a highly misleading estimate of normative change at the average age of the sample. For these reasons, we prespecified the inclusion criterion that the *SD* of age in the sample was not larger than 5 years at the first measurement occasion to ensure sufficient homogeneity regarding age. (e) The sample included at least 30 participants (to ensure sufficiently reliable sample means and *SD*s; for a similar procedure, see Orth, Erol, & Luciano, 2018). (f) The sample was not a clinical sample (see above). (g) The sample did not undergo any kind of intervention or treatment as part of the study (information from control groups without any alternative intervention/treatment was used). (h) Enough information was reported to compute effect sizes (i.e., mean and *SD* of age, mean and *SD* of SWB measure, sample size).

If two or more studies relied on the same data, we included the study with the largest sample size and the most comprehensive information on study characteristics tested as moderators. In cases where the information required for inclusion of the study was missing (e.g., mean and *SD* of SWB measure), but the study, in general, appeared to meet the inclusion criteria, the authors of the study were contacted via email. In total, we contacted 176 authors. Of those, 23 authors responded that they still had access to the data and provided us with the information necessary for inclusion. In addition to the 23 authors who provided data that could be included in this meta-analysis, another 26 authors replied, but their data could not be included for several reasons (e.g., the data were not available any longer, no longitudinal assessment of well-being was included, or information on age was not collected in the study).

Some studies have made significant contributions to examining the longitudinal development of SWB but could not be included in this meta-analysis. For example, the samples used in Jackson and Chen (2008a) as well as Jackson and Chen (2008b), which were excluded, were also used in Chen and Jackson (2009), which was included. Moreover, the samples used in Kolosnitsyna et al. (2017) and Shankar et al. (2015) could not be included in the meta-analysis because of strong age heterogeneity (i.e., a *SD* of age larger than 5 years; see the exclusion criteria described above).

Coding Procedure

The coding was performed by experienced researchers based on a prespecified coding manual (see Open Science Framework [OSF]: https://osf.io/wn7vm/; Buecker et al., 2023). We coded the following information: year of publication, publication type, sample size, sample type, the proportion of women, country of origin, ethnicity, year of Time 1 measurement, SWB component, SWB scale, the time frame of SWB measure, mean age at Time 1, SD of age at Time 1, time lag between measurement occasions, and effect size information (i.e., mean and SD of the SWB measure). If the year of Time 1 measurement was not reported in the article or other sources of information on the sample, we estimated it using the following formula: year of Time 1 measurement = publication year -3 years (assuming that studies were published on average 3 years after the completion of data collection) - interval between first and last measurement occasion (i.e., duration of data collection; for a similar procedure, see Orth, Erol, & Luciano, 2018).

For studies that included more than two measurement occasions, we coded all available occasions if they were at least 6 months apart. Moreover, if studies provided more than one effect size because two different measures of SWB were used, all measures were included. Consequently, we included multiple effect size estimates per study for some studies (see below for details on how we handled the statistical dependency between the effect sizes). If a study included more than two measurement occasions with time lags smaller than 6 months (e.g., first measurement occasion in January, the next in March, and then again in August), we used only those measurement occasions that provided information on consecutive (i.e., nonoverlapping) time intervals that were 6 months or longer (e.g., January and August).

To ensure high reliability of the codings, three coders (two postdoctoral researchers and one doctoral student) double-coded studies using a rotation principle. The basic procedure of this rotation principle was that Coder 1 independently coded 50 studies that were already coded by Coder 2. Coder 2 independently coded 50 studies that were already coded by Coder 3. Coder 3 independently coded 50 studies that were already coded by Coder 3. Coder 1. Regarding inclusion/exclusion, the intercoder agreement was 97% across all pairs of coders. Regarding the study characteristics, intercoder agreement was on average 89% across all pairs of coders (ranging from 50% to 100%). Divergent coding was discussed until the coders reached a consensus.

Computation and Analysis of Effect Sizes

All analyses were conducted in R Version 4.0.3 (R Core Team, 2020). We used the single-group, pre–post effect size to study the mean-level change in SWB across the life span (Morris & DeShon, 2002). A similar procedure has been used in previous meta-analyses on mean-level change across the life span (e.g., Luhmann, Hofmann, et al., 2012; Mund et al., 2020; Orth, Erol, & Luciano, 2018; Roberts et al., 2006). Following Morris and DeShon (2002), we calculated the standardized mean change d as follows:

$$d_i = \frac{\text{raw mean at time}_t - \text{raw mean at time}_{t-1}}{\text{standard deviation at time}_{t-1}}.$$
 (1)

A positive *d* value indicates an increase in SWB, and a negative *d* value indicates a decrease. By using these procedures (i.e., by computing the effect sizes based on means and *SD*s of the variables), we ensured that none of the effect sizes included in the meta-analytic data set had been adjusted for covariates. Thus, even if some of the original studies from which the meta-analytic data were drawn controlled for covariates in the analyses, the effect sizes of the present meta-analysis were not adjusted for any covariates. In the next step, we set the *d* in relation to the observed time interval by dividing it by the length of the time lag (in years) between time t - 1 and time *t*. Thus, our final effect size measure in the present meta-analysis is a change-to-time ratio, with the unit *d* per year.

When using standardized mean change d per year as effect size, the within-study variance per year is given by

 v_i

$$=\frac{\frac{2(1-r_i)}{n_i}+\frac{d_i^2}{2n_i}}{(\text{time lag between time}_{t-1} \text{ and time}_t)^2},$$
 (2)

where d_i is the standardized mean change d in study i, n_i is the sample size in study i, and r_i is the correlation between the measures at time t - 1 and time t in study i. The denominator in the formula is

important because we needed the within-study variance *per year* that corresponds to the effect size *d* per year (Viechtbauer, 2019). We used the test–retest correlation between time t - 1 and time *t* reported in the original study whenever possible. In cases in which r_i was not reported in the original study, we used the mean correlation across SWB components and studies (i.e., r = .523). We estimated random-effects metaregression models to generalize the findings beyond the included studies in the present meta-analysis (Raudenbush, 2009).

For the main analyses, we were interested in the *d* per year within different age groups across the life span. To create these age groups, we used the mean age of the sample at the center of the time interval on which the effect size (i.e., d per year) was based rather than the mean age of the sample at time t or time t - 1. For example, if the mean age of a sample was 12 years at time t - 1 and 16 years at time t, the age at the center of the interval on which the effect size was based was 14. We argue that this procedure is especially beneficial for studies using larger time intervals between assessments (e.g., 5 or 10 years). It would have been ideal if all studies used relatively short time intervals between assessments. Though, as the number of studies with very broad time intervals was small in this metaanalytic data set, effect sizes could be mapped on age with sufficient precision. Despite these considerations, the determination of age groups is to some extent subjective. Although using a different set of age groups might result in slightly different trajectories, the general pattern of the trajectories would be unaltered and can accordingly be considered robust. We further elaborate on this topic in the Discussion section.

For each outcome measure (i.e., life satisfaction, positive affect, and negative affect), we created age groups that were narrow enough to reflect different developmental periods but broad enough to include at least five effect sizes. We created the following 12 age groups for each outcome: 9-12 years, 12-14 years, 14-16 years, 16-18 years, 18-22 years, 22-30 years, 30-40 years, 40-50 years, 50-60 years, 60-70 years, 70-80 years, 80 to the maximum age in years. We first estimated the overall *d* per year for each age group. These *d* per year effects were then cumulated across the life span.

Accounting for the Dependency of Effect Sizes

Generally, univariate meta-analytic techniques assume that effect sizes are independent (Moeyaert et al., 2017). Yet, studies with three or more waves of data, studies with multiple samples, and studies that use different measurement scales for the same construct produce multiple effect sizes within a study, resulting in dependency of effect sizes. Previous meta-analyses of longitudinal studies often solved the dependency of effect sizes by averaging effect sizes within samples before the meta-analytic computations (e.g., Orth, Erol, & Luciano, 2018). Still, this procedure loses information and makes the meta-analytic computations less precise. Moreover, there might also be a dependency between studies that this averaging procedure cannot handle. For example, study results from the same research group can be more similar than those from different research groups, or studies from the same country can be more similar than studies from different countries. There are many different ways to deal with the issue of dependent effect sizes in meta-analyses (for an overview, see Moeyaert et al., 2017). In the present meta-analysis, we applied robust variance estimation (RVE). The RVE approach adjusts the standard errors and does not require accurate knowledge of the within-study covariance structure (Tanner-Smith & Tipton, 2014). Therefore, this approach can be applied to any type of dependency, to any degree of dependency, and to any type of effect size. Consequently, a dependency that arises from multiple measurement scales used, multiple samples within a study, or multiple measurement occasions can be accommodated simultaneously using RVE (Tanner-Smith & Tipton, 2014). Therefore, we included all available measurement points in the present metaanalysis without averaging them while taking different dependencies into account.

Moderator Analyses

We estimated a series of mixed-effects metaregressions to test for moderators of mean-level change in SWB. As for some age groups, there were only a few studies, and because some moderators were sparsely distributed across age groups, we combined all age groups in the moderator analyses. Thus, in these analyses, we examined the full sample of studies covering the observed age range from 9 to 96 years (for life satisfaction), 9 to 94 years (for positive affect), and 9 to 87 years (for negative affect). In these models, we controlled for the mean age of the sample as a continuous variable (i.e., controlling for the linear effect of age; for a similar procedure, see Orth, Erol, & Luciano, 2018). To examine the effect of birth cohort, we computed the mean year of birth using the variable's mean age at Time 1 and year of Time 1 assessment. The mean year of birth was included as a continuous variable in our moderator analyses. Moreover, we included the percentage of women in the sample as a continuous variable in our moderator analyses. For all categorical variables (i.e., publication type, country of origin of the sample, ethnicity of the sample, sample type, and time frame of SWB measures), we focused on specific contrasts due to the low number of samples in some categories. We describe these contrasts in more detail in the Results section.

Outlier Analyses

We tested for outliers and influential cases in the effect sizes because they may affect the validity and robustness of our metaanalytic conclusions (Viechtbauer & Cheung, 2010). We performed these analyses separately for each SWB outcome and across all age groups using the *influence.rma.uni* function from the *metafor* package (Viechtbauer, 2010). No dependency of effect sizes could be considered in these analyses. We excluded effect sizes if the externally standardized residuals were greater than 1.96 or smaller than -1.96 and if the effect size was considered influential. Information on when an effect size is considered influential is provided by Viechtbauer and Cheung (2010).

Publication Bias Analyses

Studies with significant findings are more likely to be published than studies with nonsignificant findings (Dickerson, 2005), a phenomenon called publication bias. It is possible that studies with samples from certain age groups (e.g., older adults) are less likely to be published if they find certain effects (e.g., increases in SWB) that contradict theoretical assumptions. In this case, publication bias would be specific to an age group. An advantage of meta-analytic techniques is the possibility to assess the extent to which the aggregated effects are affected by publication bias. To examine if there is evidence of publication bias, we tested the funnel plot asymmetry using Egger's regression test (Egger et al., 1997) separately for each age group and outcome. This analysis was performed after the exclusion of potential outliers. The basic idea of Egger's regression test is that a statistically significant association between the observed effect size and the standard error implies asymmetry in the funnel plot, which may indicate publication bias. In these analyses, we handled the dependency of effect sizes by using a multilevel metaregression with effect sizes nested within samples (i.e., by using the rma.mv function from the metafor package in R; Viechtbauer, 2010). Moreover, we tested whether the effect sizes differed depending on whether the means and SDs of the SWB outcomes were extracted from published journal articles (48%) versus any sort of gray literature (including dissertations, book chapters, unpublished data sets; 52%). The large proportion of unpublished effect sizes is due to including SWB information from multiple age groups (i.e., multiple independent samples) with many measurement occasions that have not been published in this form.

Results

Study Characteristics

For life satisfaction, we included 363 unique samples with 415,423 participants with an average sample size of N = 1,144 (*SD* = 6,031.42, *Mdn* = 282). For the positive affect, we included 128 unique samples with 88,162 participants with an average sample size of N = 688.77 (*SD* = 3,354.59, *Mdn* = 270.5). For the negative affect, we included 129 unique samples with 65,274 participants with an average sample size of N = 506 (*SD* = 709.11, *Mdn* = 247).

Supplemental Table S1 shows the sample characteristics of the included studies. Across components of SWB, females and males were on average almost equally represented in the samples (M =55% females, SD = 31, Mdn = 53). Most of the samples were from Europe (56%), followed by North America (United States and Canada; 23%). The remaining 21% came from other countries (e.g., China, Brazil, Australia). A total of 61% were community samples, 6% were college/university student samples, and 32% were nationally representative samples. The remaining 1% used other sample types (e.g., schoolchildren, sample type not specified). Of all included samples, 54% were predominantly White/European sample, 8% were predominantly Asian, 2% were predominantly Native American, 1% was predominantly Black, and 1% was predominantly Hispanic/Latin American. The rest of the samples were described as mixed ethnicity (19%), or no ethnicity information was provided for the sample (15%). Across components of SWB, the first measurement occasion of the included studies was between 1975 and 2020 (M = 2,006.63, SD = 7.99, Mdn = 2,008). Mean year of birth ranged from 1,907.16 to 2,005 (M = 1,970.21, SD = 25.39).

As described above, we included samples that had a maximum *SD* of age of 5. We chose this criterion to ensure that effect sizes can be mapped with sufficient precision on age (for a similar procedure, see Orth, Erol, & Luciano, 2018). Across all included samples, the average *SD* of age was 1.36 years. The meta-analytic data set can be assessed at OSF (https://osf.io/wn7vm/).

Preliminary Analyses

For each outcome, we investigated whether the distribution of effect sizes suggested that there were influential outliers. We excluded 15 effect sizes for life satisfaction, five for positive affect, and five for negative affect because they were identified as influential outliers. Nonetheless, as a robustness check, we also estimated mean effect sizes by retaining all effect sizes in the metaanalytic data set (see Supplemental Table S2 and Figure S1). The results suggested that the overall pattern of findings was almost the same, regardless of whether influential outliers were excluded or included.

Next, we assessed whether there was evidence of publication bias in the different age groups analyzed in this meta-analysis. We did not expect publication bias to be a major issue in this meta-analysis because most studies included did not focus on the development of SWB but simply reported the relevant descriptive statistics (i.e., means and *SD* of life satisfaction, positive affect, or negative affect). To test for publication bias, we used three different approaches.

First, we visually inspected the funnel plots, which display the association between the observed effect size and the standard error. Most funnel plots exhibited a symmetrical shape typical of nonbiased meta-analytic data sets (see Supplemental Figure S2– Figure S7).

Second, Egger's regression tests (Sterne & Egger, 2005) were conducted. Because of the large number of tests (12 age groups for each of the three outcomes, resulting in 36 tests in total), we adjusted the significance level to .001, following the Bonferroni method (i.e., dividing .05 by 36). The results of Egger's regression tests are shown in Table 1. Overall, the findings suggested that in most age groups, the funnel plots of the outcomes did not deviate significantly from a symmetrical shape, speaking against publication bias.

Third, we tested whether effect sizes from unpublished data (including dissertations) differed significantly from effect sizes from articles published in peer-reviewed journals. This analysis was performed across all age groups, because for some age groups, the amount of gray literature was too low to interpret this comparison. The unstandardized results of a mixed-effects metaregression model indicated that effect sizes from unpublished data (coded as 0) and journal articles (coded as 1) did not significantly differ for any of the outcomes (life satisfaction: B = -0.013, p = .148; positive affect: B = -0.007, p = .758; negative affect: B = 0.040, p = .073). In sum, the different publication bias analyses suggested that there is little evidence of publication bias.

Effect Size Analyses

The main aim of this meta-analysis was to map mean-level change in life satisfaction, positive affect, and negative affect on age. Therefore, we conducted the effect size analyses within age groups. Table 2 provides an overview of the age groups across the observed age range.

The relatively narrow age groups were created to maximize precision in estimating age-dependent mean-level changes. Although the power of significance tests of mean-level changes would be greater if we created broader age groups with a larger number of samples, we emphasize that null hypothesis significance tests of mean-level changes were not a central goal of this metaanalysis (Cumming, 2014; Fraley & Marks, 2007; Greenwald,

 Table 1
 Egger's Regression Test of Funnel Plot Asymmetry

	=		
Outcome	Age group	z	р
Life satisfaction	9-12 years	1.16	.247
	12-14 years	-0.15	.882
	14-16 years	2.51	.012
	16-18 years	-0.33	.740
	18-22 years	0.62	.533
	22-30 years	0.61	.539
	30-40 years	-0.87	.386
	40-50 years	-0.35	.727
	50-60 years	1.40	.161
	60-70 years	1.39	.165
	70-80 years	0.25	.804
	80–96 years	-0.11	.913
Positive affect	9–12 years	0.08	.939
	12–14 years	0.74	.458
	14–16 years	1.16	.244
	16–18 years	-0.53	.593
	18-22 years	-0.16	.869
	22-30 years	-0.78	.437
	30-40 years	-0.15	.880
	40-50 years	0.07	.942
	50-60 years	0.52	.604
	60–70 years	1.70	.089
	70-80 years	-0.59	.553
	80–94 years	-0.99	.320
Negative affect	9–12 years	-0.72	.468
	12–14 years	-0.60	.546
	14–16 years	-0.11	.910
	16–18 years	1.21	.227
	18-22 years	-0.27	.785
	22-30 years	-1.21	.227
	30-40 years	-3.04	.002
	40-50 years	-1.51	.130
	50-60 years	-0.94	.347
	60-70 years	0.28	.781
	70-80 years	1.37	.171
	80-87 years	4.39	<.001 ^a

Note. Computations were made using mixed-effects metaregression models.

^a Indicates statistical significance based on the Bonferroni-adjusted significance level of .001.

1975). For completeness, Table 2 also includes 95% confidence intervals of mean effect sizes.

Figure 2 illustrates the trajectories of life satisfaction, positive affect, and negative affect, on the basis of the meta-analytic effect sizes reported in Table 2. Given that the age groups covered more than 1 year, the estimate of yearly change (e.g., d = -0.141 for life satisfaction in the age group 9-12 years) was used for each year included in the group (e.g., 3 years for the age group 9-12, resulting in a cumulative change of d = -0.423 from age 9 to 12 years). Using these procedures, we plotted the trajectories across the observed age range for the three constructs of SWB. Life satisfaction (Figure 2A), on average, decreased from age 9 until age 16 (d = -0.56). From age 16 onward, life satisfaction increased slightly until age 70 (d = 0.16) and then decreased again (d = -0.24), which corresponds to a small effect size based on Cohen (1992). More recent guidelines for interpreting effect sizes criticize Cohen's conventions as too conservative (Funder & Ozer, 2019). Consequently, effects of such size may also be considered moderate.

Positive affect (Figure 2B) decreased, on average, from age 9 to age 70 (d = -1.64). From age 70 to age 80, positive affect increased (d = 0.20). This increase was followed by a decrease until age 94 (d = -0.27). Overall, the difference between age 9 and 94 corresponds to a large effect of d = -1.71 based on Cohen (1992).

Negative affect (Figure 2C) showed ups and downs between childhood and early adulthood. It decreased from age 9 to age 12 (d = -0.12), followed by an increase from age 12 to age 22 (d = 0.20). After age 22, negative affect decreased until age 60 (d = -0.92), corresponding to a large effect based on Cohen (1992), after which it increased again until age 87. The difference between age 60 and 87 corresponds to a medium effect of d = 0.58 based on Cohen (1992). On average, negative affect reached the lowest point at age 60 years, with a cumulative d = -1.00 (i.e., relative to age 9, which was the first age for which data on negative affect were available).

Moderator Analyses

We used mixed-effects metaregression models to test whether certain study characteristics are significant moderators of the effect sizes. For categorical variables, we focused on specific contrasts because the number of effect sizes in some of the categories was too low to draw meaningful conclusions. More specifically, for sample type, we contrasted effect sizes from nationally representative samples with effect sizes from other samples. For country, we contrasted effect sizes from European samples with effect sizes from other samples and with effect sizes from North American samples. For ethnicity, we contrasted effect sizes from White/ European samples with effect sizes from other samples. For the type of SWB measure, we contrasted effect sizes based on versions of the Satisfaction With Life Scale (Diener et al., 1985) with effect sizes based on other measures (for life satisfaction). Regarding positive and negative affect, we contrasted effect sizes based on versions of the PANAS questionnaire (Watson et al., 1988) with effect sizes based on other affect measures. Moreover, we tested whether effect sizes based on PANAS versions measuring trait-like affect (i.e., the time frame "in general" was the reference category in these contrasts) differed from effect sizes based on PANAS versions with more short-term time frames (e.g., "last few months", "last few weeks", "momentary", "past month", "past week", or "past year"). The frequencies of the different time frames used to measure positive and negative affect are presented in Supplemental Table S5. As metric moderators, we included the percentage of females, age, and year of birth. We grand-mean centered age and year of birth. Because of the large number of tests in the moderator analyses (i.e., seven tests for each outcome plus one additional test of the time frame for positive affect and negative affect each, resulting in 23 tests), we adjusted the significance level to .002 (i.e., dividing .05 by 23).

In the first step of the moderator analyses, we tested each moderator separately, controlling for the mean age of the sample. We report these results in the Supplemental Material (Supplemental Table S3 and Table S6). In the second step, we tested all moderators simultaneously, again controlling for age, to examine whether the moderators had unique effects. Table 3 reports the results of this second step. We found no significant moderating effects for any of the outcomes using the adjusted significance level of .002, suggesting that the pattern of findings was robust across studies with different sample characteristics. Moreover, the nonsignificant

	Number of	Number of			Heterog	geneity
Age group	studies	effect sizes	d per year	95% CI	I^2	τ^2
Life satisfaction						
9-12 years	7	17	-0.141	[-0.271, -0.010]	73.205	0.016
12–14 years	31	48	-0.131	[-0.174, -0.087]	94.120	0.013
14-16 years	36	65	-0.008	[-0.041, 0.026]	89.958	0.006
16–18 years	30	45	0.002	[-0.032, 0.036]	86.587	0.002
18-22 years	40	77	-0.004	[-0.027, 0.019]	84.889	0.001
22-30 years	52	142	0.004	[-0.013, 0.021]	86.760	0.001
30-40 years	49	149	0.007	[-0.007, 0.021]	98,903	0.004
40-50 years	38	117	-0.003	[-0.035, 0.029]	95.360	0.002
50–60 years	38	124	0.007	[-0.005, 0.019]	83,100	0.001
60-70 years	32	111	0.004	[-0.010, 0.017]	80.584	0.001
70-80 years	37	112	-0.013	[-0.039, 0.013]	90.994	0.001
80-96 years	29	107	-0.007	[-0.028, 0.015]	82.285	0.001
Positive affect		107	01007	[0.020, 0.010]	02.200	0.001
9–12 years	9	16	-0.068	[-0.190, 0.055]	88 451	0.043
12-14 years	22	33	-0.059	[-0.118, 0.001]	93,156	0.022
14-16 years	17	29	-0.042	[-0.105, 0.020]	74.967	0.009
16-18 years	7	7	-0.057	[-0.172, 0.057]	25.085	0.003
18-22 years	13	22	-0.044	[-0.117, 0.029]	93,400	0.011
22-30 years	11	27	-0.034	[-0.118, 0.050]	81.486	0.007
30-40 years	11	35	-0.043	[-0.096, 0.011]	80 590	0.004
40-50 years	15	41	-0.013	[-0.029, 0.003]	61.030	0.000
50-60 years	16	40	-0.007	[-0.020, 0.007]	38 379	0.000
60-70 years	9	24	-0.011	[-0.061, 0.039]	53 802	0.001
70-80 years	11	24	0.019	[-0.035, 0.074]	94.131	0.004
80-94 years	10	17	-0.019	[-0.064, 0.025]	85.556	0.002
Negative affect	10	17	01017	[0.000.1, 0.0020]	001000	0.002
9–12 years	15	26	-0.059	[-0.176, 0.057]	93.547	0.029
12-14 years	27	38	0.029	[-0.018, 0.075]	94.833	0.014
14-16 years	24	40	0.035	[-0.007, 0.076]	74 208	0.005
16-18 years	13	17	0.012	[-0.180, 0.203]	93,920	0.035
18-22 years	14	31	0.012	[-0.064, 0.090]	98.655	0.034
22-30 years	11	36	-0.042	[-0.150, 0.065]	84.347	0.010
30-40 years	8	33	-0.044	[-0.083, -0.005]	58.057	0.003
40-50 years	8	33	-0.022	[-0.065, 0.021]	68.581	0.001
50-60 years	12	41	-0.009	[-0.061, 0.043]	78.745	0.005
60-70 years	8	30	0.014	[-0.028, 0.056]	56.143	0.002
70-80 years	10	31	0.025	[-0.043, 0.093]	58.061	0.002
80–87 years	6	25	0.027	[-0.019, 0.073]	63.233	0.001

 Table 2

 Meta-Analytic Results for Life Satisfaction, Positive Affect, and Negative Affect in Each Age Group

Note. CI = confidence interval.

effect of mean year of birth indicates no evidence of cohort effects. For each moderator (except time frame due to the large number of categories), we also tested whether it interacted with the mean age of the sample in predicting the effect sizes. This analysis was included because the effects of some variables (e.g., gender) on SWB might be more pronounced in certain phases in life (i.e., at a certain age) than in others. Nevertheless, none of the interaction effects were statistically significant (see Supplemental Table S4).

Discussion

The present meta-analysis aimed to draw a precise and comprehensive picture of the normative trajectories of SWB (as indicated by life satisfaction, positive affect, and negative affect) across the life span. Our meta-analytic data set included 443 unique samples with a total of 460,902 participants across all outcomes. Overall, the mean-level trajectories differed substantially among the three components of SWB. Life satisfaction decreased from age 9 to 16 (d = -0.56), increased slightly until age 70 (d = 0.16), and then decreased again until age 96 (i.e., the oldest age for which data on life satisfaction were available; d = -0.24). Positive affect declined from childhood for almost the entire time until age 94 (d = -1.71). Negative affect showed smaller ups and downs between age 9 and 22. After age 22, negative affect declined until age 60 (d = -0.92), after which it increased again until age 87 (d = 0.58). Average changes in positive and negative affect were more pronounced than in life satisfaction. Overall, the maximum of cumulative change for life satisfaction corresponds to a moderate effect (d = -0.56), whereas the maximum of cumulative change for positive affect (d =-1.71) and negative affect (d = -1.00) corresponds to a large effect based on Cohen (1992). A similar effect size (d > 1) has also been found for self-esteem and the Big Five traits of emotional stability and openness (Bleidorn et al., 2022; Orth, Erol, & Luciano, 2018).

Figure 2 *Mean-Level Change in Life Satisfaction (A), Positive Affect (B), and Negative Affect (C) Across the Life Span*



Note. The vertical axis shows cumulative d values relative to age 9 (thus, the point of origin [i.e., zero] is arbitrary; for a similar way of illustrating meta-analytic findings on mean-level change, see Orth, Erol, & Luciano, 2018 and Roberts et al., 2006). For age groups that covered more than 1 year (e.g., ages 18–22), the estimate of the yearly change was used for each year included in the age group.

Development of SWB Across the Life Span

Most theoretical perspectives reviewed in the introduction predict that SWB changes in normative ways across the life span. Only the set point theory suggests that SWB is stable across the life span, an idea that is clearly challenged by our meta-analytic results. In this section, we discuss the meta-analytic findings separately for the different phases of life. We discuss the findings against the background of the hypotheses derived from theoretical perspectives relevant to these phases of life.

Childhood and Adolescence

The disruption hypothesis (Brandes et al., 2021; Soto & Tackett, 2015) suggested a temporary decrease in life satisfaction and positive affect and an increase in negative affect in the period from

Table 3	
Moderator Analyse	s

Outcome	Estimate	В	SE	р
Life satisfaction $(k = 662)$	Intercept	-0.02	0.03	.457
	Age at T1	-0.00	0.00	.682
	Year of birth	-0.00	0.00	.341
	Sample type (representative)	0.01	0.01	.257
	Country (North America)	0.00	0.02	.933
	Country (other)	-0.02	0.03	.420
	Ethnicity (White/European)	0.01	0.02	.731
	Percentage of females	-0.00	0.00	.967
	Type of SWB measure (SWLS)	0.00	0.01	.774
Positive affect $(k = 148)$	Intercept	-0.01	0.05	.838
	Age at T1	0.00	0.00	.960
	Year of birth	0.00	0.00	.447
	Sample type (representative)	-0.02	0.02	.184
	Country (North America)	-0.05	0.04	.197
	Country (other)	0.03	0.04	.572
	Ethnicity (White/European)	-0.01	0.04	.788
	Percentage of females	0.00	0.00	.790
	Type of SWB measure (PANAS)	0.00	0.02	.950
Negative affect $(k = 256)$	Intercept	0.03	0.07	.696
	Age at T1	0.00	0.00	.625
	Year of birth	0.00	0.00	.440
	Sample type (representative)	0.04	0.03	.191
	Country (North America)	-0.07	0.06	.265
	Country (other)	-0.11	0.10	.280
	Ethnicity (White/European)	-0.05	0.06	.410
	Percentage of females	0.00	0.00	.829
	Type of SWB measure (PANAS)	0.03	0.03	.380

Note. Computations were made using mixed-effects metaregression models, in which all predictors were added simultaneously. Mean age was included as a control variable in all models. Mean age and mean year of birth were grand-mean centered. The following variables were dichotomous: sample type (1 = representative, 0 = other), ethnicity (1 = White/European, 0 = other), type of SWB measure (1 = SWLS [for life satisfaction] or PANAS [for positive and negative affect], 0 = other). For country, we used two dummy variables representing three categories of the country variable (Europe vs. North America vs. other countries). The reference category was Europe. *k* = the number of effect sizes, *B* = unstandardized regression coefficient from the mixed-effects metaregression model, including study characteristics as predictors of effect sizes. *SE* = standard error; SWB = subjective well-being; T1 = Time 1; PANAS = Positive and Negative Affect Scale; SWLS = Satisfaction With Life Scale.

middle childhood to late adolescence (i.e., between age 9 and age 18). For life satisfaction, our meta-analytic findings are consistent with this hypothesis. For positive and negative affect, however, the interpretation is less clear. Positive affect decreased from childhood on. Still, our meta-analytic results do not suggest that this decrease in positive affect is a disruption (i.e., a temporary dip) because the decreasing trend continued until late adulthood. In contrast, negative affect tended to increase from age 12 until age 22. After that age, the negative affect decreased. Thus, this finding is consistent with the idea of disruption during adolescence and early adulthood. Across all three outcomes, the findings suggest that SWB worsens during adolescence, consistent with the disruption hypothesis. Yet, this deterioration does not appear as temporary for all outcomes as suggested by the disruption hypothesis.

Several mechanisms could be responsible for this developmental trend. For example, the transition from childhood to adulthood is characterized by different physical, psychological, and social changes (Goldbeck et al., 2007), which may lead to a decrease in SWB in adolescence. Neurobiological theories suggest that age-related changes in the morphological structure of the brain (Giedd, 2004) and in the neurotransmitter system (Wahlstrom et al., 2010) lead to losses in SWB

during adolescence. Moreover, certain developmental tasks such as moving toward independence (Geuzaine et al., 2000) may confront adolescents with emotional challenges that result in decreases in SWB. These emotional challenges fall into a period of life where the repertoire of emotion regulation strategies is still relatively small, which is a risk factor for low SWB (Zimmermann & Iwanski, 2014). Our meta-analytic findings are also in line with previous crosssectional findings suggesting lower levels of life satisfaction and quality of life during adolescence (Goldbeck et al., 2007).

Moreover, the disruption hypothesis suggests greater losses in SWB for adolescent girls than for adolescent boys (Brandes et al., 2021; Soto & Tackett, 2015). In our meta-analysis, we did not find significant gender effects. In this regard, it should be noted that in the moderator analyses, we could only test for the effect of the average proportion of females or males in a sample. Thus, in this meta-analysis, the nonsignificant moderator effect refers to differences between samples, which does not necessarily generalize to differences between individuals. In future research, it would be worthwhile to test whether the nonsignificant gender effect holds when tested with individual-level data on the development of SWB.

Young Adulthood

For young adulthood (i.e., age 18–40 years), the theoretical perspectives led to conflicting hypotheses about the development of SWB. The maturity principle outlined in the neo-socioanalytic theory (Roberts & Robins, 2021; Roberts & Wood, 2006) suggests a decrease in negative affect and an increase in positive affect and life satisfaction in young adulthood. When considering normative changes and transitions during young adulthood (e.g., moving out, beginning of higher education or vocational training, starting a family), both losses and gains in SWB are plausible.

Regarding young adulthood, the meta-analytic findings do not clearly support one of the theoretical perspectives reviewed in the introduction. Regarding life satisfaction, we found an increase in line with the maturity principle and with studies showing that young adulthood is characterized by a high density of positive life events and life circumstances (Lüdtke et al., 2011). Yet, the average increase in life satisfaction was small. Regarding the affective component of SWB, we found that both positive and negative affect decreased during young adulthood.

At first sight, it might be surprising that positive and negative affect both decrease and that there is no complementary developmental course (e.g., a decrease in positive affect, accompanied by an increase in negative affect, or vice versa). Nonetheless, research has long shown that positive and negative affect are-although related-not opposite poles of the same construct but must be distinguished conceptually and empirically (Diener & Emmons, 1984). As described in the introduction, the general affect intensity (for positive and negative affect) seems to decrease from young adulthood until late adulthood (Larsen & Diener, 1987). Moreover, the present findings could also be explained by the ambivalence of changes in life circumstances during young adulthood. On the one hand, young adulthood is often described as the rush hour of life (Zannella et al., 2019), in which multiple and partly conflicting tasks need to be accomplished. These tasks typically involve making choices, for example, with regard to one's education, career path, romantic partner, and whether or not to start a family. The simultaneous demands in different life domains can be perceived as stressful, resulting in decreases in positive affect. On the other hand, although making choices can be stressful, this period in life is characterized by high perceived control, which is generally linked to lower negative affect (Drewelies et al., 2018). Additionally, the financial situation starts to stabilize in this developmental period (Bea & Yi, 2019; Knudson & Mazurik, 2021), and most young adults do not yet suffer from any major health problems. Taken together, young adulthood could be described as a period with high demands and possibly decreasing positive affect, but also with high resources (e.g., financial, social, physical, and psychological functioning) and possibly decreasing negative affect, which is what we found in this metaanalysis. The decreasing trend in both positive and negative affect is also in line with the idea of decreasing affect intensity across the life span (Bailen et al., 2019).

Middle Adulthood

For middle adulthood (i.e., age 40–65 years), neo-socioanalytic theory suggested an increase in SWB due to further maturation of people's personality traits. Consistent with this hypothesis, life satisfaction increased in middle adulthood, whereas negative affect

decreased in this period. These findings may also be understood within the theory of dynamic integration (Labouvie-Vief, 2003), which postulates that the balance between the optimization of positive emotions and experiences and tolerance of negative emotions stabilizes in middle adulthood. Positive affect, however, decreased in middle adulthood, which is inconsistent with the prediction from neo-socioanalytic theory and the theory of dynamic integration. A possible explanation is that the size of people's social networks—which is associated with SWB (e.g., Litwin & Levinsky, 2022)—steadily becomes smaller in middle adulthood (Wrzus, Hänel, et al., 2013).

Late Adulthood

For late adulthood (i.e., 65 years and older), theoretical perspectives again proposed conflicting hypotheses. Neosocioanalytic theory (Roberts & Robins, 2021; Roberts & Wood, 2006) and the terminal decline hypothesis (Gerstorf et al., 2008; Gerstorf & Ram, 2015) suggest a decrease in SWB in late adulthood. Similarly, the theory of dynamic integration (Labouvie-Vief, 2003) postulates that aging is associated with difficulties in integrating and tolerating negative affect, which may result in increases in negative affect. In contrast, socioemotional selectivity theory suggests an increase in SWB due to emotion regulation strategies that help the individual to select positive situations more efficiently and to compensate for losses. Moreover, the SAVI model (Charles, 2010) adds a differential perspective on changes in SWB during late adulthood by emphasizing that the SWB trajectory may strongly depend on one's age-related strengths and vulnerabilities. As late adulthood is, however, on average associated with situations in which strengths cannot be exploited, decreases in well-being can be assumed based on the SAVI model.

The present meta-analytic findings show that life satisfaction and positive affect decrease in late adulthood, whereas negative affect increases. Thus, these findings are consistent with the hypotheses from the neo-socioanalytic theory, the theory of dynamic integration, the SAVI model, and the terminal decline hypothesis and inconsistent with the hypothesis from the socioemotional selectivity theory. A possible explanation is that older adults (compared to young or middle-aged adults) typically experience more losses in important life domains such as health (Furman et al., 2019; Wilson et al., 2020) and social relationships (Buecker et al., 2021; Luhmann, Hofmann, et al., 2012). According to the SAVI model (Charles, 2010), the vulnerabilities outweigh the strengths. Furthermore, numerous studies have suggested that negative age stereotypes and perceived age discrimination may interfere with older adults' SWB (Kornadt & Rothermund, 2011; Kotter-Grühn & Hess, 2012; Wurm et al., 2017).

Is There a U-Shaped Trajectory of SWB?

As reviewed in the introduction, researchers have debated whether SWB follows a U-shaped trajectory over the life course, with the nadir in middle adulthood (Blanchflower & Graham, 2021a; Galambos et al., 2020, 2021). A U-shape would be consistent with the idea that people experience, at least on average, some kind of midlife crisis, where SWB is the lowest. The present metaanalysis does not support the hypothesis of a U-shape, for neither life satisfaction, nor positive affect, nor negative affect. Specifically, life satisfaction generally increased from age 16 to age 70, with a slight downturn between age 40 and age 50 (across these 10 years, the aggregated decrease corresponded to d = -0.03). The very small effect size of the midlife decrease in life satisfaction does not correspond to the hypothesis of a U-shaped trajectory of SWB, with a significant low point in midlife. Also, positive affect continuously decreased from age 9 to age 94, which again is inconsistent with the U-shape hypothesis. Finally, negative affect was lowest in midlife, which is the opposite of what would be expected on the basis of the U-shape hypothesis for SWB, and which is inconsistent with the idea of a midlife crisis. Thus, although many laypeople intuitively agree with the concept of the midlife crisis and although many people report that they have experienced a midlife crisis (Freund & Ritter, 2009), this subjective assessment is inconsistent with the longitudinal trajectory as determined in this meta-analysis. Still, even if the present research suggests no normative low point in SWB in midlife, the notion of a midlife crisis might refer to other experiences in people's lives. For example, a recent meta-analysis suggests that people's satisfaction with their romantic relationship shows the lowest point at age 40 (Bühler et al., 2021).

Moderator Results

We expected heterogeneity in SWB trajectories because SWB levels are determined by multiple factors, including personal characteristics (e.g., personality traits, genetic/biological factors, psychological resources), external context (e.g., general life circumstances, life events, geographical/cultural/historical context), and activities/behaviors (e.g., how people spend their everyday lives). In theory, changes in any of these factors might lead to changes in SWB. Since these factors do not change in unison for everyone, heterogeneity of SWB trajectories can be expected. We performed moderator analyses to test whether this heterogeneity could be explained by the variables coded in the meta-analysis.

These moderator analyses indicated that none of the moderators examined explained variability in the effect sizes. The nonsignificant moderator of the birth cohort appears to be particularly noteworthy. The mean year of birth ranged from 1907 to 2005 across the samples included in this meta-analysis. Consequently, the nonsignificant cohort effect indicates that the shape of the life span trajectory of SWB has not changed over the generations born during the 20th century when examining longitudinal studies. Such nonsignificant cohort effects are not uncommon in metaanalyses of longitudinal studies. For example, Bleidorn et al. (2022) also found no cohort effects in the development of the Big Five personality traits. Even still, this nonsignificant finding does not necessarily mean that there is no cohort effect in studies using other research designs (e.g., in cross-sectional data, developmental trends and cohort effects are confounded). Although samples that differed in their mean year of birth typically also differed in their mean age, we argue that our moderator analyses provide valid insights into cohort effects (i.e., because the analyses statistically controlled for the effect of age). Thus, the effect of birth year captured the unique cohort effect while holding age constant (and also while controlling for other variables, such as ethnicity or sample type, which otherwise could have confounded the effect).

Moreover, the nonsignificant cohort effect is required for valid interpretations of life span trajectories based on estimates across different birth cohorts (as done in this meta-analysis). For example, if more recent generations had experienced steeper decreases in life satisfaction during late adulthood compared to previous generations, these cohort differences would have confounded the overarching life span trajectory of life satisfaction, leading to invalid conclusions. Consequently, the nonsignificant cohort effects in this meta-analysis (for life satisfaction, positive affect, and negative affect) support the validity of the conclusions about the SWB trajectories. Importantly, this meta-analysis provides information only about cohort effects on mean-level *change* in SWB (i.e., cohort effects on the slope) but not the mean level of SWB itself. Thus, future research might examine whether more recent generations of people of certain ages show higher or lower levels of SWB.

The other nonsignificant moderator effects indicate that our overall conclusions regarding the life span trajectories of life satisfaction, positive affect, and negative affect can be considered robust across different samples and study characteristics. Nonetheless, this finding does not imply that there are no moderators but only that no significant evidence of moderators was found, based on the available meta-analytic data set (which was large and included several hundred studies).

Some theoretical perspectives focus on change in momentary, state-like SWB (e.g., affect regulation), while others draw on change in habitual SWB (e.g., maturation principle). A direct test of whether trajectories in SWB are different for state-like or trait-like SWB could not be performed in this meta-analysis. Instead, we examined the time frame of the measurement instruments as a moderator and did not find any evidence for a moderation effect. Still, it should be noted that the time frame does not always resolve whether SWB is state-like or trait-like (Robinson & Clore, 2002). For example, it has not yet been clearly clarified empirically whether a recall of one's own SWB in the past weeks refers more to state-like or trait-like aspects of SWB (compared to asking for SWB "in general" which more clearly refers to trait-like SWB).

Implications of the Present Findings for the Structure of SWB

In a landmark article, Diener (1984) had introduced the tripartite model of SWB, proposing that SWB consists of life satisfaction, positive affect, and negative affect and that these three constructs reflect distinct dimensions. Even though this model is very influential, it has been criticized by others as "premature" (Busseri & Sadava, 2011, p. 1). Specifically, Busseri and colleagues have argued that life satisfaction, positive affect, and negative affect might not be distinct constructs but rather reflect a hierarchical construct with SWB as a higher order factor and life satisfaction, positive affect, and negative affect as only partially independent lower order factors (Busseri, 2015; Busseri & Sadava, 2011). Clearly, the present metaanalysis did not examine the dimensional structure of SWB. Nonetheless, the meta-analytic findings indicate that, at least regarding the normative trajectory of SWB, it is crucial to distinguish between life satisfaction, positive affect, and negative affect. In fact, for each construct, the meta-analysis suggested a distinct developmental pattern across the life span, which speaks for the conceptual distinction of the three SWB components and for the need to assess each of the constructs separately. We note that consistent with the results of the present meta-analysis, Busseri (2015) found that life satisfaction shows greater mean-level stability over time than positive affect and negative affect.

Our meta-analytical findings showing that life satisfaction was more stable (or more precisely, changed more slowly) than positive and negative affect across the life span are also in line with metaanalytic evidence on changes in these components surrounding major life events (Luhmann, Hofmann, et al., 2012). Life satisfaction is typically conceptualized as people's overall evaluation of their global life circumstances. In contrast, affective well-being (i.e., positive and negative affect) is conceptualized as people's evaluations of recent activities or events, which are more transient than global life circumstances (Luhmann, Hawkley, et al., 2012). Consequently, it seems plausible that life satisfaction and affect differ in their temporal stability across the life span. One might argue that this difference is artificially enforced by differences in measuring the cognitive and affective domains of SWB (e.g., different time frames are used in the scales). Yet, multitrait-multimethod analyses indicated that life satisfaction and affective well-being were empirically distinct even if the same time frame in the measurement was used (Luhmann, Hawkley, et al., 2012), suggesting more substantive than methodological differences between the SWB components. The nonsignificant moderator effect of the time frame in this meta-analysis also supports this perspective.

Limitations

For all SWB outcomes, the pattern of findings was robust across several samples and study characteristics. This finding strengthens confidence in the generalizability of our results. Moreover, we used an analytical approach ensuring that—even if original studies from which the meta-analytic data were drawn had adjusted for covariates in their analyses—the effect sizes of the present metaanalysis were not adjusted for any covariates. Despite these strengths, some limitations still exist that need to be discussed.

First, although we included published and unpublished records from 32 countries and collected information on ethnicity, country of origin, and gender, one limitation of this meta-analysis is that most samples were from Western countries (e.g., Europe, North America) with predominantly White/European samples. Consequently, we were unable to test whether effect sizes from Asian or South American samples differed from effect sizes from Western countries. In line with the recommendations by Johnson (2021), we explicitly point out limitations regarding diversity in longitudinal studies on SWB and call for future research with samples from non-Western countries and more diverse samples in terms of ethnicity. We encourage studying the development of SWB across the life span in such samples and examining whether cultural characteristics may be a source of heterogeneity in trajectories of SWB or whether the trajectories reported in this meta-analysis replicate across different cultures. We emphasize that the trajectories of the various SWB components may also vary as a function of interindividual differences within cultures. To test the moderating effect of such interindividual differences on SWB trajectories, more longitudinal studies collecting data on these variables will be needed in the future.

Second, in this meta-analysis, we created 12 age groups for each outcome, estimated the average mean-level change in these age groups, and cumulated these changes to examine the trajectories across the life span. In creating the age groups, we pursued two goals: First, the age groups should be narrow enough to be able to represent theoretically relevant changes (e.g., during puberty). Second, the age groups should contain a sufficient number of effect sizes so that the mean effect size for this age group can be estimated with sufficient precision. Because we assumed complex nonlinear trajectories of SWB, we decided against modeling age as a continuous predictor in a metaregression. When life span trajectories are nonlinear, metaregression cannot capture the complex and subtle changes that are present in the data (e.g., the ups and downs of negative affect in adolescence).

Third, in a meta-analysis, one can estimate only the *average* trajectory across the life span and thus gain a broad overview of a field of research. The effects found in this meta-analysis were heterogeneous and average trajectories do not necessarily apply to every individual. We conducted several moderator analyses to explain this heterogeneity. None of the moderators tested were statistically significant, indicating the robustness of our findings. Even still, we could not test other potentially relevant moderators such as the socioeconomic status or the relationship status of the sample. Moreover, in a meta-analysis, moderator tests are performed on the level of samples (e.g., the proportion of females in a sample) not on the level of individuals (e.g., individual gender). It remains a task for future research to examine further moderators on the trajectory of SWB across the life span.

Finally, we would like to stress that the mean trajectories of the three SWB components reported in this meta-analysis (as shown in Figure 2) are based on cumulative effect sizes. For each of the estimated effect sizes (i.e., within age groups), the analyses provided a 95% confidence interval indicating the degree of uncertainty in the estimate (Table 2). It was not possible to compute confidence intervals for the cumulative effect sizes, although clearly there is some degree of uncertainty in the trajectories shown in Figure 2. The interpretation of these meta-analytic trajectories should, therefore, be done with appropriate caution. For example, even if Figure 2 shows a small increase or decrease in the trajectory in a given age period (e.g., the slight increase in life satisfaction from young to middle adulthood), the true trajectory could be flat rather than an increase or decrease. Nevertheless, given that the goal of the present research was to track the normative trajectory of SWB across the life span (by synthesizing the available longitudinal information), the point estimates of the effect sizes represent the best estimates for describing this average trajectory.

Conclusions

Based on longitudinal data from 443 samples with more than 460,000 participants, this meta-analysis provides a comprehensive overview of normative mean-level changes in life satisfaction, positive affect, and negative affect across the life span. We found that life satisfaction decreased during adolescence, increased in young and middle adulthood, and decreased again in late adulthood. Positive affect decreased almost the entire time until late adulthood. Negative affect showed ups and downs from age 9 to 22, decreased during young and middle adulthood, and

increased in late adulthood. Taken together, the results of the present meta-analysis suggest that the three components of SWB show distinct developmental patterns across the life span. When generalizing across the SWB components, we found a favorable developmental trajectory of SWB over large parts of life (i.e., from adolescence until about 70 years) for life satisfaction and negative affect but not for positive affect, which decreased over large parts of life. Beginning at age 70 years, the general trend became more negative and SWB worsened rather than improved. Consequently, the findings suggest that interventions aimed at maintaining or enhancing SWB might be particularly useful in late adulthood. Given that the evidence on the effectiveness of wellbeing interventions is still sparse (for a systematic review, see Owen et al., 2021), future research in this area is strongly needed.

References

References marked with an asterisk indicate studies included in the metaanalysis.

- *Abele, A. E., Hagmaier, T., & Spurk, D. (2016). Does career success make you happy? The mediating role of multiple subjective success evaluations. *Journal of Happiness Studies: An Interdisciplinary Forum on Subjective Well-Being*, 17(4), 1615–1633. https://doi.org/10.1007/s10902-015-9662-4
- *Adamczyk, K. (2017). Going beyond relationship status: A cross-sectional and longitudinal investigation of the role of satisfaction with relationship status in predicting Polish young adults' mental health. *Journal of Social* and Clinical Psychology, 36(4), 265–284. https://doi.org/10.1521/jscp .2017.36.4.265
- *Adie, J. W., Duda, J. L., & Ntoumanis, N. (2010). Achievement goals, competition appraisals, and the well- and ill-being of elite youth soccer players over two competitive seasons. *Journal of Sport & Exercise Psychology*, 32(4), 555–579. https://doi.org/10.1123/jsep.32.4.555
- *Allan, B. A., Owens, R. L., Kim, T., Douglass, R. P., & Hintz, J. (2019). Strengths and satisfaction in first year undergraduate students: A longitudinal study. *The Journal of Positive Psychology*, *16*(1), 94–104. https://doi.org/10.1080/17439760.2019.1676458
- Alwin, D. F., & Wray, L. A. (2005). A life-span developmental perspective on social status and health. *The Journals of Gerontology. Series B, Psychological Sciences and Social Sciences*, 60(2), 7–14. https://doi.org/ 10.1093/geronb/60.Special_Issue_2.S7
- *Anglim, J., Weinberg, M. K., & Cummins, R. A. (2015). Bayesian hierarchical modeling of the temporal dynamics of subjective well-being: A 10 year longitudinal analysis. *Journal of Research in Personality*, 59, 1–14. https://doi.org/10.1016/j.jrp.2015.08.003
- *Armenta, B. E., Sittner Hartshorn, K. J., Whitbeck, L. B., Crawford, D. M., & Hoyt, D. R. (2014). A longitudinal examination of the measurement properties and predictive utility of the Center for Epidemiologic Studies Depression Scale among North American Indigenous adolescents. *Psychological Assessment*, 26(4), 1347–1355. https://doi.org/10.1037/a0037608
- Baetschmann, G. (2014). Heterogeneity in the relationship between happiness and age: Evidence from the German socio-economic panel. *German Economic Review*, 15(3), 393–410. https://doi.org/10.1111/geer.12015
- Bailen, N. H., Green, L. M., & Thompson, R. J. (2019). Understanding emotion in adolescents: A review of emotional frequency, intensity, instability, and clarity. *Emotion Review*, 11(1), 63–73. https://doi.org/10 .1177/1754073918768878
- Baird, B. M., Lucas, R. E., & Donnellan, M. B. (2010). Life satisfaction across the lifespan: Findings from two nationally representative panel studies. *Social Indicators Research*, 99(2), 183–203. https://doi.org/10 .1007/s11205-010-9584-9

- Baltes, P. B., Cornelius, S. W., & Nesselroade, J. R. (1979). Cohort effects in developmental psychology. In J. R. Nesselroade & P. B. Baltes (Eds.), *Longitudinal research in the study of behavior and development* (pp. 61– 87). Academic Press.
- *Bares, C. B., & Andrade, F. H. (2012). Racial/ethnic differences in the longitudinal progression of co-occurring negative affect and cigarette use: From adolescence to young adulthood. *Addictive Behaviors*, 37(5), 632–640. https://doi.org/10.1016/j.addbeh.2012.01.016
- *Barker, E. T., Howard, A. L., Galambos, N. L., & Wrosch, C. (2016). Tracking affect and academic success across university: Happy students benefit from bouts of negative mood. *Developmental Psychology*, 52(12), 2022–2030. https://doi.org/10.1037/dev0000231
- Bartels, M. (2015). Genetics of wellbeing and its components satisfaction with life, happiness, and quality of life: A review and meta-analysis of heritability studies. *Behavior Genetics*, 45(2), 137–156. https://doi.org/ 10.1007/s10519-015-9713-y
- Bartram, D. (2021). Age and life satisfaction: Getting control variables under control. *Sociology*, 55(2), 421–437. https://doi.org/10.1177/00380 38520926871
- Bauer, J. M., Levin, V., Munoz Boudet, A. M., Nie, P., & Sousa-Poza, A. (2017). Subjective well-being across the lifespan in Europe and central Asia. *Journal of Population Ageing*, 10(2), 125–158. https://doi.org/10 .1007/s12062-016-9148-0
- Bea, M. D., & Yi, Y. (2019). Leaving the financial nest: Connecting young adults' financial independence to financial security. *Journal of Marriage* and Family, 81(2), 397–414. https://doi.org/10.1111/jomf.12553
- *Bearman, S. K., Martinez, E., Stice, E., & Presnell, K. (2006). The skinny on body dissatisfaction: A longitudinal study of adolescent girls and boys. *Journal of Youth and Adolescence*, 35(2), 217–229. https://doi.org/10 .1007/s10964-005-9010-9
- *Berg, A. I., Hassing, L. B., Thorvaldsson, V., & Johansson, B. (2011). Personality and personal control make a difference for life satisfaction in the oldest-old: Findings in a longitudinal population-based study of individuals 80 and older. *European Journal of Ageing*, 8(1), 13–20. https://doi.org/10.1007/s10433-011-0181-9
- Berg, A. I., Hoffman, L., Hassing, L. B., McClearn, G. E., & Johansson, B. (2009). What matters, and what matters most, for change in life satisfaction in the oldest-old? A study over 6 years among individuals 80+. *Aging & Mental Health*, 13(2), 191–201. https://doi.org/10.1080/13607860802342227
- *Bergman, L. R., & Daukantaite, D. (2009). Stability of typical patterns of subjective well-being in middle-aged Swedish women. *Journal* of Happiness Studies, 10(3), 293–311. https://doi.org/10.1007/s10902-007-9081-2
- *Bergold, S., & Steinmayr, R. (2023). Teacher judgments predict adolescents' development in school performance, motivation, and life satisfaction [Unpublished manuscript]. Institute of Psychology, Technical University of Dortmund.
- Biermann, P., Bitzer, J., & Gören, E. (2022). The relationship between age and subjective well-being: Estimating within and between effects simultaneously. *The Journal of the Economics of Ageing*, 21, Article 100366. https://doi.org/10.1016/j.jeoa.2021.100366
- Blanchflower, D. G., & Graham, C. L. (2021a). The U-shape of happiness: A response. *Perspectives on Psychological Science*, 16(6), 1435–1446. https://doi.org/10.1177/1745691620984393
- Blanchflower, D. G., & Graham, C. L. (2021b). The mid-life dip in wellbeing: A critique. Social Indicators Research, 161, 287–344. https:// doi.org/10.1007/s11205-021-02773-w
- Blanchflower, D. G., & Oswald, A. J. (2008). Is well-being U-shaped over the life cycle? *Social Science & Medicine*, 66(8), 1733–1749. https:// doi.org/10.1016/j.socscimed.2008.01.030
- Bleidorn, W. (2015). What accounts for personality maturation in early adulthood? *Current Directions in Psychological Science*, 24(3), 245–252. https://doi.org/10.1177/0963721414568662

- Bleidorn, W., Kandler, C., Riemann, R., Spinath, F. M., & Angleitner, A. (2009). Patterns and sources of adult personality development: Growth curve analyses of the NEO PI-R scales in a longitudinal twin study. *Journal of Personality and Social Psychology*, 97(1), 142–155. https:// doi.org/10.1037/a0015434
- Bleidorn, W., Klimstra, T. A., Denissen, J. J. A., Rentfrow, P. J., Potter, J., & Gosling, S. D. (2013). Personality maturation around the world: A crosscultural examination of social-investment theory. *Psychological Science*, 24(12), 2530–2540. https://doi.org/10.1177/0956797613498396
- Bleidorn, W., Schwaba, T., Zheng, A., Hopwood, C. J., Sosa, S. S., Roberts, B. W., & Briley, D. A. (2022). Personality stability and change: A meta-analysis of longitudinal studies. *Psychological Bulletin*, 148(7–8), 588–619. https://doi.org/10.1037/bul0000365
- *Boer, M., Stevens, G. W. J. M., Finkenauer, C., & van den Eijnden, R. J. J. M. (2022). The course of problematic social media use in young adolescents: A latent class growth analysis. *Child Development*, 93(2), e168–e187. https://doi.org/10.1111/cdev.13712
- Boisvert, S., & Poulin, F. (2016). Romantic relationship patterns from adolescence to emerging adulthood: Associations with family and peer experiences in early adolescence. *Journal of Youth and Adolescence*, 45(5), 945–958. https://doi.org/10.1007/s10964-016-0435-0
- *Borghuis, J., Bleidorn, W., Sijtsma, K., Branje, S., Meeus, W. H. J., & Denissen, J. J. A. (2020). Longitudinal associations between trait neuroticism and negative daily experiences in adolescence. *Journal of Personality and Social Psychology*, 118(2), 348–363. https://doi.org/10.1037/pspp0000233
- Bradburn, N. M. (1969). The structure of psychological well-being. Aldine Publishing.
- Braithwaite, S. R., Delevi, R., & Fincham, F. D. (2010). Romantic relationships and the physical and mental health of college students. *Personal Relationships*, 17(1), 1–12. https://doi.org/10.1111/j.1475-6811 .2010.01248.x
- Brandes, C. M., Kushner, S. C., Herzhoff, K., & Tackett, J. L. (2021). Facet-level personality development in the transition to adolescence: Maturity, disruption, and gender differences. *Journal of Personality and Social Psychology*, 121(5), 1095–1111. https://doi.org/10.1037/pspp0000367
- *Bratt, C. (2015). One of few or one of many: Social identification and psychological well-being among minority youth. *British Journal of Social Psychology*, 54(4), 671–694. https://doi.org/10.1111/bjso.12105
- Brickman, P., & Campbell, D. (1971). Adaptation-level theory: A symposium. In M. H. Appley (Ed.), *Hedonic relativism and planning the good society* (pp. 287–305). Academic Press.
- Brose, A., Lindenberger, U., & Schmiedek, F. (2013). Affective states contribute to trait reports of affective well-being. *Emotion*, 13(5), 940– 948. https://doi.org/10.1037/a0032401
- Buecker, S., Denissen, J. J. A., & Luhmann, M. (2021). A propensity-score matched study of changes in loneliness surrounding major life events. *Journal of Personality and Social Psychology*, 121(3), 669–690. https:// doi.org/10.1037/pspp0000373
- Buecker, S., Luhmann, M., Haehner, P., Buehler, J. L., Dapp, L. C., Luciano, E. C., & Orth, U. (2023, May 31). The development of subjective well-being across the lifespan: A meta-analytic review of longitudinal studies. https://osf.io/wn7vm
- Bühler, J. L., Krauss, S., & Orth, U. (2021). Development of relationship satisfaction across the life span: A systematic review and meta-analysis. *Psychological Bulletin*, 147(10), 1012–1053. https://doi.org/10.1037/bu 10000342
- Bühler, J. L., & Nikitin, J. (2020). Sociohistorical context and adult social development: New directions for 21st century research. *American Psychologist*, 75(4), 457–469. https://doi.org/10.1037/amp0000611
- *Bühler, J. L., Weidmann, R., Wünsche, J., Burriss, R. P., & Grob, A. (2020). Daily responsiveness, expectations, and self-disclosure: How the average levels and within-person variability of three relationship components mediate personality-relationship transactions in romantic

couples. European Journal of Personality, 34(3), 367-392. https://doi.org/ 10.1002/per.2255

- *Burger, K., & Samuel, R. (2017). The role of perceived stress and selfefficacy in young people's life satisfaction: A longitudinal study. *Journal* of Youth and Adolescence, 46(1), 78–90. https://doi.org/10.1007/s10964-016-0608-x
- *Burr, A., Santo, J. B., & Pushkar, D. (2011). Affective well-being in retirement: The influence of values, money, and health across three years. *Journal of Happiness Studies*, 12(1), 17–40. https://doi.org/10.1007/ s10902-009-9173-2
- *Burris, J. L., Riley, E., Puleo, G. E., & Smith, G. T. (2017). A longitudinal study of the reciprocal relationship between ever smoking and urgency in early adolescence. *Drug and Alcohol Dependence*, *178*, 519–526. https://doi.org/10.1016/j.drugalcdep.2017.06.007
- Busseri, M. A. (2015). Toward a resolution of the tripartite structure of subjective well-being: Structure of subjective well-being. *Journal of Personality*, 83(4), 413–428. https://doi.org/10.1111/jopy.12116
- *Busseri, M. A., Choma, B. L., & Sadava, S. W. (2009). Functional or fantasy? Examining the implications of subjective temporal perspective "trajectories" for life satisfaction. *Personality and Social Psychology Bulletin*, 35(3), 295–308. https://doi.org/10.1177/0146167208327215
- Busseri, M. A., & Sadava, S. W. (2011). A review of the tripartite structure of subjective well-being: Implications for conceptualization, operationalization, analysis, and synthesis. *Personality and Social Psychology Review*, 15(3), 290–314. https://doi.org/10.1177/1088868310391271
- *Busseri, M. A., & Sadava, S. W. (2013). Subjective well-being as a dynamic and agentic system: Evidence from a longitudinal study. *Journal of Happiness Studies*, 14(4), 1085–1112. https://doi.org/10.1007/s10902-012-9368-9
- *Cappeliez, P., & Robitaille, A. (2010). Coping mediates the relationships between reminiscence and psychological well-being among older adults. *Aging & Mental Health*, 14(7), 807–818. https://doi.org/10.1080/ 13607861003713307
- *Carmel, S. (2017). The will-to-live scale: Development, validation, and significance for elderly people. *Aging & Mental Health*, 21(3), 289–296. https://doi.org/10.1080/13607863.2015.1081149
- *Carmel, S., Raveis, V. H., O'Rourke, N., & Tovel, H. (2017). Health, coping and subjective well-being: Results of a longitudinal study of elderly Israelis. *Aging & Mental Health*, 21(6), 616–623. https://doi.org/ 10.1080/13607863.2016.1141285
- Carstensen, L. L., Isaacowitz, D. M., & Charles, S. T. (1999). Taking time seriously. A theory of socioemotional selectivity. *American Psychologist*, 54(3), 165–181. https://doi.org/10.1037/0003-066X.54.3.165
- *Casas, F., & González, M. (2022). Do relationships between subjective well-being scales change over time? Analysis of a longitudinal sample. *Current Psychology: A Journal for Diverse Perspectives on Diverse Psychological Issues*, 41(6), 3525–3538. https://doi.org/10.1007/s12144-020-00865-3
- Caspi, A., Roberts, B. W., & Shiner, R. L. (2005). Personality development: Stability and change. *Annual Review of Psychology*, 56(1), 453–484. https://doi.org/10.1146/annurev.psych.55.090902.141913
- *Cebotari, V., Mazzucato, V., & Appiah, E. (2018). A longitudinal analysis of well-being of Ghanaian children in transnational families. *Child Development*, 89(5), 1768–1785. https://doi.org/10.1111/cdev.12879
- *CentERdata. (2022). LISS (longitudinal internet studies for the social sciences) panel [Unpublished raw data]. Centerdata. https://www.lissda ta.nl
- *Chao, S. (2010). Life transitions, social support and psychological wellbeing among the elderly in Taiwan: A longitudinal study. University at Albany.
- Charles, S. T. (2010). Strength and vulnerability integration: A model of emotional well-being across adulthood. *Psychological Bulletin*, 136(6), 1068–1091. https://doi.org/10.1037/a0021232

- *Chen, H., & Jackson, T. (2009). Predictors of changes in body image concerns of Chinese adolescents. *Journal of Adolescence*, 32(4), 977–994. https://doi.org/10.1016/j.adolescence.2008.08.002
- *Chen, H. Y., & Cheng, C. L. (2020). Developmental trajectory of purpose identification during adolescence: Links to life satisfaction and depressive symptoms. *Journal of Adolescence*, 80(1), 10–18. https://doi.org/10.1016/ j.adolescence.2020.01.013
- *Chen, H.-Y., Chiou, H., & Cheng, C.-L. (2022). Purpose trajectories during middle adolescence: The roles of family, teacher, and peer support. *Journal of Youth and Adolescence*, 51(2), 291–304. https://doi.org/10 .1007/s10964-021-01548-3
- *Chen, K.-M. (2020). Subjective poverty, deprivation, and the subjective well-being of children and young people: A multilevel growth curve analysis in Taiwan. *Children and Youth Services Review*, 114, Article 105045. https://doi.org/10.1016/j.childyouth.2020.105045
- Cheng, T. C., Powdthavee, N., & Oswald, A. J. (2017). Longitudinal evidence for a midlife nadir in human well-being: Results from four data sets. *Economic Journal*, 127(599), 126–142. https://doi.org/10.1111/ecoj.12256
- Cheung, F., & Lucas, R. E. (2015). When does money matter most? Examining the association between income and life satisfaction over the life course. *Psychology and Aging*, *30*(1), 120–135. https://doi.org/10 .1037/a0038682
- Cohen, J. (1992). A power primer. *Psychological Bulletin*, *112*(1), 155–159. https://doi.org/10.1037/0033-2909.112.1.155
- Costa, P. T., & McCrae, R. R. (1982). An approach to the attribution of aging, period, and cohort effects. *Psychological Bulletin*, 92(1), 238–250. https://doi.org/10.1037/0033-2909.92.1.238
- *Cowan, H. R. (2019). Can a good life be unsatisfying? Within-person dynamics of life satisfaction and psychological well-being in late midlife. *Psychological Science*, 30(5), 697–710. https://doi.org/10.1177/095679 7619831981
- *Cowlishaw, S., Niele, S., Teshuva, K., Browning, C., & Kendig, H. (2013). Older adults' spirituality and life satisfaction: A longitudinal test of social support and sense of coherence as mediating mechanisms. *Ageing and Society*, 33(7), 1243–1262. https://doi.org/10.1017/S0144686X12000633
- Cumming, G. (2014). The new statistics: Why and how. *Psychological Science*, 25(1), 7–29. https://doi.org/10.1177/0956797613504966
- *Dainese, S. M., Allemand, M., Ribeiro, N., Bayram, S., Martin, M., & Ehlert, U. (2011). Protective factors in midlife: How do people stay healthy? *GeroPsych: The Journal of Gerontopsychology and Geriatric Psychiatry*, 24(1), 19–29. https://doi.org/10.1024/1662-9647/a000032
- *Daniel, E., Andersen, J. P., & Papazoglou, K. (2016). Social identification and well-being following a terrorist attack: A longitudinal study of Israeli adolescents. *The Journal of Genetic Psychology: Research and Theory on Human Development*, 177(5), 172–184. https://doi.org/10.1080/00221325 .2016.1223010
- *Daukantaitė, D., & Zukauskiene, R. (2012). Optimism and subjective wellbeing: Affectivity plays a secondary role in the relationship between optimism and global life satisfaction in the middle-aged women. Longitudinal and cross-cultural findings. *Journal of Happiness Studies*, *13*(1), 1–16. https://doi.org/10.1007/s10902-010-9246-2
- *Davis, H. A., Ortiz, A. M. L., & Smith, G. T. (2019). Transactions between early binge eating and personality predict transdiagnostic risk. *European Eating Disorders Review*, 27(6), 614–627. https://doi.org/10.1002/ erv.2682
- *Delfabbro, P., Winefield, H., Winefield, A., Malvaso, C., & Plueckhahn, T. (2017). Factors associated with attrition in a 10-year longitudinal study of young people: Implications for studies of employment in school leavers. *Australian Psychologist*, 52(1), 41–51. https://doi.org/10.1111/ap.12207
- DeNeve, K. M., & Cooper, H. (1998). The happy personality: A metaanalysis of 137 personality traits and subjective well-being. *Psychological Bulletin*, 124(2), 197–229. https://doi.org/10.1037/0033-2909.124.2.197

- Denissen, J. J. A., Luhmann, M., Chung, J. M., Bleidorn, W., & Chung, J. M. (2019). Transactions between life events and personality traits across the adult lifespan. *Journal of Personality and Social Psychology*, *116*(4), 612–633. https://doi.org/10.1037/pspp0000196
- Denissen, J. J. A., van Aken, M. A. G., Penke, L., & Wood, D. (2013). Selfregulation underlies temperament and personality: An integrative developmental framework. *Child Development Perspectives*, 7(4), 255– 260. https://doi.org/10.1111/cdep.12050
- Dickerson, K. (2005). Publication bias: Recognizing the problem, understandings its origins and scope, and preventing harm. In H. R. Rothstein, A. J. Sutton, & M. Borenstein (Eds.), *Publication bias in meta analysis: Prevention, assessment, and adjustments* (pp. 11–34). Wiley. https:// doi.org/10.1002/0470870168.ch2
- Diener, E. (1984). Subjective well-being. *Psychological Bulletin*, 95(3), 542–575. https://doi.org/10.1037/0033-2909.95.3.542
- Diener, E., & Chan, M. Y. (2011). Happy people live longer: Subjective well-being contributes to health and longevity. *Applied Psychology*. *Health and Well-Being*, 3(1), 1–43. https://doi.org/10.1111/j.1758-0854 .2010.01045.x
- Diener, E., & Emmons, R. A. (1984). The independence of positive and negative affect. *Journal of Personality and Social Psychology*, 47(5), 1105–1117. https://doi.org/10.1037/0022-3514.47.5.1105
- Diener, E., Emmons, R. A., Larsen, R. J., & Griffin, S. (1985). The Satisfaction With Life Scale. *Journal of Personality Assessment*, 49(1), 71–75. https://doi.org/10.1207/s15327752jpa4901_13
- Diener, E., Lucas, R. E., & Scollon, C. N. (2006). Beyond the hedonic treadmill: Revising the adaptation theory of well-being. *American Psychologist*, 61(4), 305–314. https://doi.org/10.1037/0003-066X.61.4.305
- Diener, E., Oishi, S., & Lucas, R. E. (2015). National accounts of subjective well-being. *American Psychologist*, 70(3), 234–242. https://doi.org/10 .1037/a0038899
- Diener, E., Suh, E. M., Lucas, R. E., & Smith, H. L. (1999). Subjective wellbeing: Three decades of progress. *Psychological Bulletin*, 125(2), 276– 302. https://doi.org/10.1037/0033-2909.125.2.276
- *Dill, E. J., Vernberg, E. M., Fonagy, P., Twemlow, S. W., & Gamm, B. K. (2004). Negative affect in victimized children: The roles of social withdrawal, peer rejection, and attitudes toward bullying. *Journal of Abnormal Child Psychology*, 32(2), 159–173. https://doi.org/10.1023/B: JACP.0000019768.31348.81
- Dolan, E., & Sale, C. (2019). Protein and bone health across the lifespan. *The Proceedings of the Nutrition Society*, 78(1), 45–55. https://doi.org/ 10.1017/S0029665118001180
- Drewelies, J., Huxhold, O., & Gerstorf, D. (2019). The role of historical change for adult development and aging: Towards a theoretical framework about the how and the why. *Psychology and Aging*, *34*(8), 1021–1039. https://doi.org/10.1037/pag0000423
- Drewelies, J., Schade, H., Hülür, G., Hoppmann, C. A., Ram, N., & Gerstorf, D. (2018). The more we are in control, the merrier? Partner perceived control and negative affect in the daily lives of older couples. *The Journals* of Gerontology: Series B, 75(2), 338–348. https://doi.org/10.1093/gero nb/gby009
- *Duffy, R. D., Manuel, R. S., Borges, N. J., & Bott, E. M. (2011). Calling, vocational development, and well being: A longitudinal study of medical students. *Journal of Vocational Behavior*, 79(2), 361–366. https://doi.org/ 10.1016/j.jvb.2011.03.023
- *Duineveld, J. J., Parker, P. D., Ryan, R. M., Ciarrochi, J., & Salmela-Aro, K. (2017). The link between perceived maternal and paternal autonomy support and adolescent well-being across three major educational transitions. *Developmental Psychology*, 53(10), 1978–1994. https:// doi.org/10.1037/dev0000364
- Dush, C. M. K., & Amato, P. R. (2005). Consequences of relationship status and quality for subjective well-being. *Journal of Social and*

Personal Relationships, 22(5), 607-627. https://doi.org/10.1177/0265 407505056438

- Egger, M., Davey Smith, G., Schneider, M., & Minder, C. (1997). Bias in meta-analysis detected by a simple, graphical test. *The BMJ*, 315(7109), 629–634. https://doi.org/10.1136/bmj.315.7109.629
- *Eisman, A. B., Stoddard, S. A., Bauermeister, J. A., Caldwell, C. H., & Zimmerman, M. A. (2017). Trajectories of organized activity participation among urban adolescents: Associations with young adult outcomes. *Journal of Community Psychology*, 45(4), 513–527. https://doi.org/10 .1002/jcop.21863
- *Ellberg, C. C., Sayler, K., & Hibel, L. C. (2020). Maternal distress across the postnatal period is associated with infant secretory immunoglobulin A. *Developmental Psychobiology*, 62(4), 544–553. https://doi.org/10.1002/ dev.21934
- *Eloranta, S., Arve, S., Lavonius, S., Routasalo, P., Lehtonen, A., Viitanen, M., & Isoaho, H. (2012). Positive life orientation in old age: A 15-year follow-up. Archives of Gerontology and Geriatrics, 55(3), 586–591. https://doi.org/10.1016/j.archger.2012.04.010
- *Esnaola, I., Benito, M., Antonio-Agirre, I., Axpe, I., & Lorenzo, M. (2019). Longitudinal measurement invariance of the Satisfaction With Life Scale in adolescence. *Quality of Life Research*, 28(10), 2831–2837. https:// doi.org/10.1007/s11136-019-02224-7
- *Fergusson, D. M., McLeod, G. F. H., Horwood, L. J., Swain, N. R., Chapple, S., & Poulton, R. (2015). Life satisfaction and mental health problems (18 to 35 years). *Psychological Medicine*, 45(11), 2427–2436. https://doi.org/10.1017/S0033291715000422
- *Fosco, G. M., & Feinberg, M. E. (2015). Cascading effects of interparental conflict in adolescence: Linking threat appraisals, self-efficacy, and adjustment. *Development and Psychopathology*, 27(1), 239–252. https:// doi.org/10.1017/S0954579414000704
- Fraley, R. C., & Marks, M. J. (2007). The null hypothesis significance-testing debate and its implications for personality research. In R. W. Robins, R. C. Fraley, & R. F. Krueger (Eds.), *Handbook of research methods in personality psychology* (pp. 149–169). Guilford.
- *Freund, A. M. (2021). *Jacobs Diet Study* [Unpublished raw data]. University of Zurich.
- *Freund, A. M., Mayer, Z., Almeling, L., & Isaacowitz, D. M. (2021). Links between motivation, emotion regulation and well-being across adulthood [Unpublished manuscript]. University of Zurich.
- *Freund, A. M., & Meyer, M. (2021). Beloved: How does caring for a dog affect successful aging? [Unpublished raw data]. University of Zurich.
- Freund, A. M., & Ritter, J. O. (2009). Midlife crisis: A debate. *Gerontology*, 55(5), 582–591. https://doi.org/10.1159/000227322
- *Frison, E., Subrahmanyam, K., & Eggermont, S. (2016). The short-term longitudinal and reciprocal relations between peer victimization on Facebook and adolescents' well-being. *Journal of Youth and Adolescence*, 45(9), 1755–1771. https://doi.org/10.1007/s10964-016-0436-z
- Funder, D. C., & Ozer, D. J. (2019). Evaluating effect size in psychological research: Sense and nonsense. Advances in Methods and Practices in Psychological Science, 2(2), 156–168. https://doi.org/10.1177/251524 5919847202
- Furman, D., Campisi, J., Verdin, E., Carrera-Bastos, P., Targ, S., Franceschi, C., Ferrucci, L., Gilroy, D. W., Fasano, A., Miller, G. W., Miller, A. H., Mantovani, A., Weyand, C. M., Barzilai, N., Goronzy, J. J., Rando, T. A., Effros, R. B., Lucia, A., Kleinstreuer, N., ... Slavich, G. M. (2019). Chronic inflammation in the etiology of disease across the life span. *Nature Medicine*, 25(12), 1822–1832. https://doi.org/10.1038/s41591-019-0675-0
- Galambos, N. L., Barker, E. T., & Krahn, H. J. (2006). Depression, self-esteem, and anger in emerging adulthood: Seven-year trajectories. *Developmental Psychology*, 42(2), 350–365. https://doi.org/10.1037/0012-1649.42.2.350
- Galambos, N. L., Krahn, H. J., Johnson, M. D., & Lachman, M. E. (2020). The U shape of happiness across the life course: Expanding the discussion.

Perspectives on Psychological Science, 15(4), 898–912. https://doi.org/ 10.1177/1745691620902428

- Galambos, N. L., Krahn, H. J., Johnson, M. D., & Lachman, M. E. (2021). Another attempt to move beyond the cross-sectional U shape of happiness: A reply. *Perspectives on Psychological Science*, *16*(6), 1447–1455. https://doi.org/10.1177/17456916211008823
- *Galla, B. M., Tsukayama, E., Park, D., Yu, A., & Duckworth, A. L. (2020). The mindful adolescent: Developmental changes in nonreactivity to inner experiences and its association with emotional well-being. *Developmental Psychology*, 56(2), 350–363. https://doi.org/10.1037/dev0000877
- *Garcia, D., & Moradi, S. (2012). Adolescents' temperament and character: A longitudinal study on happiness. *Journal of Happiness Studies*, 13(5), 931–946. https://doi.org/10.1007/s10902-011-9300-8
- *García-Peña, C., Wagner, F. A., Sánchez-García, S., Espinel-Bermúdez, C., Juárez-Cedillo, T., Pérez-Zepeda, M., Arango-Lopera, V., Franco-Marina, F., Ramírez-Aldana, R., & Gallo, J. J. (2013). Late-life depressive symptoms: Prediction models of change. *Journal of Affective Disorders*, *150*(3), 886–894. https://doi.org/10.1016/j.jad.2013.05.007
- George, L. K., Okun, M. A., & Landerman, R. (1985). Age as a moderator of the determinants of life satisfaction. *Research on Aging*, 7(2), 209–233. https://doi.org/10.1177/0164027585007002004
- *Gerber, M., Brand, S., Feldmeth, A. K., Lang, C., Elliot, C., Holsboer-Trachsler, E., & Pühse, U. (2013). Adolescents with high mental toughness adapt better to perceived stress: A longitudinal study with Swiss vocational students. *Personality and Individual Differences*, 54(7), 808–814. https://doi.org/10.1016/j.paid.2012.12.003
- Gerstorf, D., Hoppmann, C. A., Löckenhoff, C. E., Infurna, F. J., Schupp, J., Wagner, G. G., & Ram, N. (2016). Terminal decline in well-being: The role of social orientation. *Psychology and Aging*, 31(2), 149–165. https:// doi.org/10.1037/pag0000072
- Gerstorf, D., Hülür, G., Drewelies, J., Willis, S. L., Schaie, K. W., & Ram, N. (2020). Adult development and aging in historical context. *American Psychologist*, 75(4), 525–539. https://doi.org/10.1037/amp0000596
- Gerstorf, D., & Ram, N. (2015). A framework for studying mechanisms underlying terminal decline in well-being. *International Journal* of Behavioral Development, 39(3), 210–220. https://doi.org/10.1177/ 0165025414565408
- Gerstorf, D., Ram, N., Estabrook, R., Schupp, J., Wagner, G. G., & Lindenberger, U. (2008). Life satisfaction shows terminal decline in old age: Longitudinal evidence from the German Socio-Economic Panel Study (SOEP). *Developmental Psychology*, 44(4), 1148–1159. https:// doi.org/10.1037/0012-1649.44.4.1148
- Geuzaine, C., Debry, M., & Liesens, V. (2000). Separation from parents in late adolescence: The same for boys and girls? *Journal of Youth and Adolescence*, 29(1), 79–91. https://doi.org/10.1023/A:1005173205791
- Giedd, J. N. (2004). Structural magnetic resonance imaging of the adolescent brain. Annals of the New York Academy of Sciences, 1021(1), 77–85. https://doi.org/10.1196/annals.1308.009
- *Gillham, J., Adams-Deutsch, Z., Werner, J., Reivich, K., Coulter-Heindl, V., Linkins, M., Winder, B., Peterson, C., Park, N., Abenavoli, R., Contero, A., & Seligman, M. E. P. (2011). Character strengths predict subjective well-being during adolescence. *The Journal of Positive Psychology*, 6(1), 31–44. https://doi.org/10.1080/17439760.2010.536773
- *Goebel, J., Grabka, M. M., Liebig, S., Kroh, M., Richter, D., Schröder, C., & Schupp, J. (2018). The German Socio-Economic Panel (SOEP). Journal of Economics and Statistics/Jahrbucher Fur Nationalokonomie und Statistik, 239(2), 345–360. https://doi.org/10.1515/jbnst-2018-0022
- Goldbeck, L., Schmitz, T. G., Besier, T., Herschbach, P., & Henrich, G. (2007). Life satisfaction decreases during adolescence. *Quality of Life Research*, 16(6), 969–979. https://doi.org/10.1007/s11136-007-9205-5
- *Gomez-Baya, D., Mendoza, R., Gaspar, T., & Gomes, P. (2018). Responses to positive affect, life satisfaction and self-esteem: A cross-lagged panel

438

analysis during middle adolescence. *Scandinavian Journal of Psychology*, 59(4), 462–472. https://doi.org/10.1111/sjop.12450

- González-Carrasco, M., Casas, F., Malo, S., Viñas, F., & Dinisman, T. (2017). Changes with age in subjective well-being through the adolescent years: Differences by gender. *Journal of Happiness Studies*, *18*(1), 63–88. https://doi.org/10.1007/s10902-016-9717-1
- *Gray, J. S., & Ozer, D. J. (2019). Comparing two models of dyadic change: Correlated growth versus common fate. *Social Psychological & Personality Science*, 10(7), 957–965. https://doi.org/10.1177/1948550618799066
- Greenwald, A. G. (1975). Consequences of prejudice against the null hypothesis. *Psychological Bulletin*, 82(1), 1–20. https://doi.org/10.1037/h0076157
- *Griffith, J. M., Clark, H. M., Haraden, D. A., Young, J. F., & Hankin, B. L. (2021). Affective development from middle childhood to late adolescence: Trajectories of mean-level change in negative and positive affect. *Journal of Youth and Adolescence*, 50(8), 1550–1563. https:// doi.org/10.1007/s10964-021-01425-z
- *Grosz, M. P., Nagengast, B., Trautwein, U., Lechner, C., & Roberts, B. W. (2019). *Is the desire for fame related to well-being or ill-being?* [Manuscript submitted for publication]. HMU Health and Medical University. https://doi.org/10.31234/osf.io/dw6qa
- Grühn, D., Lumley, M. A., Diehl, M., & Labouvie-Vief, G. (2013). Timebased indicators of emotional complexity: Interrelations and correlates. *Emotion*, 13(2), 226–237. https://doi.org/10.1037/a0030363
- *Gustavson, K., Nilsen, W., Ørstavik, R., & Røysamb, E. (2014). Relationship quality, divorce, and well-being: Findings from a three-year longitudinal study. *The Journal of Positive Psychology*, 9(2), 163–174. https:// doi.org/10.1080/17439760.2013.858274
- *Gustavson, K., Røysamb, E., Borren, I., Torvik, F. A., & Karevold, E. (2016). Life satisfaction in close relationships: Findings from a longitudinal study. *Journal of Happiness Studies*, 17(3), 1293–1311. https://doi.org/10.1007/s10902-015-9643-7
- *Haase, C. M., Heckhausen, J., & Köller, O. (2008). Goal engagement during the school-work transition: Beneficial for all, particularly for girls. *Journal of Research on Adolescence*, 18(4), 671–698. https://doi.org/10 .1111/j.1532-7795.2008.00576.x
- *Haase, C. M., Poulin, M. J., & Heckhausen, J. (2012). Happiness as a motivator: Positive affect predicts primary control striving for career and educational goals. *Personality and Social Psychology Bulletin*, 38(8), 1093–1104. https://doi.org/10.1177/0146167212444906
- *Hackett, R. A., Steptoe, A., & Jackson, S. E. (2019). Sex discrimination and mental health in women: A prospective analysis. *Health Psychology*, 38(11), 1014–1024. https://doi.org/10.1037/hea0000796
- Hang, Y., Speyer, L. G., Murray, A. L., Luciano, M., & Mõttus, R. (2023). Social expectations as a possible mechanism for adult personality change: Limited empirical evidence for the social investment principle. *Journal of Personality*. Advance online publication. https://doi.org/10.1111/jopy.12809
- *Haraden, D. A., Mullin, B. C., & Hankin, B. L. (2019). Internalizing symptoms and chronotype in youth: A longitudinal assessment of anxiety, depression and tripartite model. *Psychiatry Research*, 272, 797–805. https://doi.org/10.1016/j.psychres.2018.12.117
- *Harding, J. F., & Sibley, C. G. (2013). The palliative function of system justification: Concurrent benefits versus longer-term costs to wellbeing. *Social Indicators Research*, 113(1), 401–418. https://doi.org/10.1007/ s11205-012-0101-1
- *Harris, K., English, T., Harms, P. D., Gross, J. J., & Jackson, J. J. (2017). Why are extraverts more satisfied? Personality, social experiences, and subjective well-being in college. *European Journal of Personality*, 31(2), 170–186. https://doi.org/10.1002/per.2101
- *Hatano, K., Hihara, S., Nakama, R., Tsuzuki, M., Mizokami, S., & Sugimura, K. (2022). Trajectories in sense of identity and relationship with life satisfaction during adolescence and young adulthood. *Developmental Psychology*, 58(5), 977–989. https://doi.org/10.1037/dev0001326

- *Hatano, K., & Sugimura, K. (2017). Is adolescence a period of identity formation for all youth? Insights from a four-wave longitudinal study of identity dynamics in Japan. *Developmental Psychology*, 53(11), 2113–2126. https://doi.org/10.1037/dev0000354
- Heckhausen, J., Wrosch, C., & Schulz, R. (2010). A motivational theory of life-span development. *Psychological Review*, 117(1), 32–60. https:// doi.org/10.1037/a0017668
- *Helgeson, V. S., Mascatelli, K., Reynolds, K. A., Becker, D., Escobar, O., & Siminerio, L. (2015). Friendship and romantic relationships among emerging adults with and without type 1 diabetes. *Journal of Pediatric Psychology*, 40(3), 359–372. https://doi.org/10.1093/jpepsy/jsu069
- *Henning, G., Bjälkebring, P., Stenling, A., Thorvaldsson, V., Johansson, B., & Lindwall, M. (2019). Changes in within- and between-person associations between basic psychological need satisfaction and well-being after retirement. *Journal of Research in Personality*, 79, 151–160. https:// doi.org/10.1016/j.jrp.2019.03.008
- *Henschel, N. T. (2019). Time stable predictors of life satisfaction in older German employees [Master's thesis, Jacobs University Bremen].
- *Horn, A. B., Pössel, P., & Hautzinger, M. (2011). Promoting adaptive emotion regulation and coping in adolescence: A school-based programme. *Journal of Health Psychology*, 16(2), 258–273. https://doi.org/10.1177/ 1359105310372814
- Howell, R. T., Kern, M. L., & Lyubomirsky, S. (2007). Health benefits: Meta-analytically determining the impact of well-being on objective health outcomes. *Health Psychology Review*, 1(1), 83–136. https://doi.org/ 10.1080/17437190701492486
- *Hsu, H.-C. (2009). Physical function trajectories, depressive symptoms, and life satisfaction among the elderly in Taiwan. *Aging & Mental Health*, *13*(2), 202–212. https://doi.org/10.1080/13607860802342201
- Hudson, N. W., Lucas, R. E., & Donnellan, M. B. (2016). Getting older, feeling less? A cross-sectional and longitudinal investigation of developmental patterns in experiential well-being. *Psychology and Aging*, 31(8), 847–861. https://doi.org/10.1037/pag0000138
- Hudson, N. W., Lucas, R. E., & Donnellan, M. B. (2019). The development of subjective well-being across the lifespan. In D. K. Mroczek & T. D. Little (Eds.), *Handbook of personality development* (pp. 503–517). Lawrence Erlbaum.
- Hutteman, R., Hennecke, M., Orth, U., Reitz, A. K., & Specht, J. (2014). Developmental tasks as a framework to study personality development in adulthood and old age. *European Journal of Personality*, 28(3), 267–278. https://doi.org/10.1002/per.1959
- *Işık Akın, R., Breeman, L. D., & Branje, S. (2021). Motivation to leave home during the transition to emerging adulthood among Turkish adolescents. *Journal of Youth Studies*, 24(10), 1273–1290. https://doi.org/ 10.1080/13676261.2020.1820970
- Jackson, T., & Chen, H. (2008a). Predicting changes in eating disorder symptoms among Chinese adolescents: A 9-month prospective study. *Journal of Psychosomatic Research*, 64(1), 87–95. https://doi.org/10 .1016/j.jpsychores.2007.08.015
- Jackson, T., & Chen, H. (2008b). Predicting changes in eating disorder symptoms among adolescents in China: An 18-month prospective study. *Journal of Clinical Child and Adolescent Psychology*, 37(4), 874–885. https://doi.org/10.1080/15374410802359841
- *Jackson, T., & Chen, H. (2014). Risk factors for disordered eating during early and middle adolescence: A two year longitudinal study of mainland Chinese boys and girls. *Journal of Abnormal Child Psychology*, 42(5), 791–802. https://doi.org/10.1007/s10802-013-9823-z
- *Jiang, Y., Ming, H., Tian, Y., Huang, S., Sun, L., Li, H., & Zhang, H. (2020). Cumulative risk and subjective well-being among rural-to-urban migrant adolescents in China: Differential moderating roles of stress mindset and resilience. *Journal of Happiness Studies*, 21(7), 2429–2449. https://doi.org/10.1007/s10902-019-00187-7

- Johnson, B. T. (2021). Toward a more transparent, rigorous, and generative psychology. *Psychological Bulletin*, 147(1), 1–15. https://doi.org/10 .1037/bul0000317
- *Jose, P. E., Ryan, N., & Pryor, J. (2012). Does social connectedness promote a greater sense of well-being in adolescence over time? *Journal* of Research on Adolescence, 22(2), 235–251. https://doi.org/10.1111/j .1532-7795.2012.00783.x
- *Jovanović, V. (2019). Measurement invariance of the Serbian version of the Satisfaction With Life Scale across age, gender, and time. *European Journal of Psychological Assessment*, 35(4), 555–563. https://doi.org/ 10.1027/1015-5759/a000410
- *Jung, S., & Choi, E. (2017). Life satisfaction and delinquent behaviors among Korean adolescents. *Personality and Individual Differences*, 104, 104–110. https://doi.org/10.1016/j.paid.2016.07.039
- *Kahana, E., Bhatta, T., Lovegreen, L. D., Kahana, B., & Midlarsky, E. (2013). Altruism, helping, and volunteering: Pathways to well-being in late life. *Journal of Aging and Health*, 25(1), 159–187. https://doi.org/10 .1177/0898264312469665
- *Kanazawa, S., Li, N. P., & Yong, J. C. (2022). When intelligence hurts and ignorance is bliss: Global pandemic as an evolutionarily novel threat to happiness. *Journal of Personality*, 90(6), 971–987. https://doi.org/10 .1111/jopy.12709
- *Kashima, E. S., Kent, S., & Kashima, Y. (2015). Life satisfaction in the new country: A multilevel longitudinal analysis of effects of culture and 5-HTT allele frequency distribution in country of origin. *Social Cognitive and Affective Neuroscience*, 10(1), 50–54. https://doi.org/10.1093/scan/ nsu036
- *Kasperzack, D., Ernst, A. L., & Pinquart, M. (2014). Ambivalence during and after career decision making of high school graduates. *Journal of Career Assessment*, 22(2), 248–260. https://doi.org/10.1177/1069072713493765
- King, L. A., & Napa, C. K. (1998). What makes a life good? Journal of Personality and Social Psychology, 75(1), 156–165. https://doi.org/10 .1037/0022-3514.75.1.156
- *King, R. B. (2017). A fixed mindset leads to negative affect: The relations between implicit theories of intelligence and subjective well-being. *Zeitschrift für Psychologie*, 225(2), 137–145. https://doi.org/10.1027/ 2151-2604/a000290
- *Kipp, L. E., & Weiss, M. R. (2015). Social predictors of psychological need satisfaction and well-being among female adolescent gymnasts: A longitudinal analysis. *Sport, Exercise, and Performance Psychology*, 4(3), 153–169. https://doi.org/10.1037/spy0000033
- *Knecht, M., & Freund, A. M. (2017). The use of selection, optimization, and compensation (SOC) in goal pursuit in the daily lives of middle-aged adults. *European Journal of Developmental Psychology*, 14(3), 350–366. https://doi.org/10.1080/17405629.2016.1207518
- Knudson, S., & Mazurik, K. (2021). Seeking stability: A preliminary exploration of Canadian young adults' financial goals. *Journal of Youth Studies*, 24(3), 285–304. https://doi.org/10.1080/13676261.2020.1716964
- *Kobayashi, E., Liang, J., Sugawara, I., Fukaya, T., Shinkai, S., & Akiyama, H. (2015). Associations between social networks and life satisfaction among older Japanese: Does birth cohort make a difference? *Psychology* and Aging, 30(4), 952–966. https://doi.org/10.1037/pag0000053
- *Kohut, T., & Štulhofer, A. (2018). Is pornography use a risk for adolescent well-being? An examination of temporal relationships in two independent panel samples. *PLOS ONE*, *13*(8), Article e0202048. https://doi.org/10 .1371/journal.pone.0202048
- *Koivumaa-Honkanen, H., Kaprio, J., Honkanen, R., Viinamäki, H., & Koskenvuo, M. (2004). Life satisfaction and depression in a 15-year follow-up of healthy adults. *Social Psychiatry and Psychiatric Epidemiology*, 39(12), 994–999. https://doi.org/10.1007/s00127-004-0833-6
- *Koivumaa-Honkanen, H., Kaprio, J., Honkanen, R. J., Viinamäki, H., & Koskenvuo, M. (2005). The stability of life satisfaction in a 15-year

follow-up of adult Finns healthy at baseline. *BMC Psychiatry*, 5(1), Article 4. https://doi.org/10.1186/1471-244X-5-4

- *Kokko, K., & Feldt, T. (2018). Longitudinal profiles of mental well-being as correlates of successful aging in middle age. *International Journal* of Behavioral Development, 42(5), 485–495. https://doi.org/10.1177/ 0165025417739177
- *Kokko, K., Tolvanen, A., & Pulkkinen, L. (2013). Associations between personality traits and psychological well-being across time in middle adulthood. *Journal of Research in Personality*, 47(6), 748–756. https:// doi.org/10.1016/j.jrp.2013.07.002
- *Koletzko, S. H., Herrmann, M., & Brandstätter, V. (2015). Unconflicted goal striving: Goal ambivalence as a mediator between goal selfconcordance and well-being. *Personality and Social Psychology Bulletin*, *41*(1), 140–156. https://doi.org/10.1177/0146167214559711
- Kolosnitsyna, M., Khorkina, N., & Dorzhiev, H. (2017). Determinants of life satisfaction in older Russians. *Ageing International*, 42(3), 354–373. https://doi.org/10.1007/s12126-017-9297-3
- Kornadt, A. E., & Rothermund, K. (2011). Contexts of aging: Assessing evaluative age stereotypes in different life domains. *The Journals of Gerontology: Series B*, 66(5), 547–556. https://doi.org/10.1093/geronb/ gbr036
- Kotter-Grühn, D., & Hess, T. M. (2012). The impact of age stereotypes on self-perceptions of aging across the adult lifespan. *The Journals of Gerontology: Series B*, 67(5), 563–571. https://doi.org/10.1093/geronb/ gbr153
- Kraemer, H. C., Yesavage, J. A., Taylor, J. L., & Kupfer, D. (2000). How can we learn about developmental processes from cross-sectional studies, or can we? *The American Journal of Psychiatry*, 157(2), 163–171. https:// doi.org/10.1176/appi.ajp.157.2.163
- *Kramer, K. Z., Bae, H., Huh, C., & Pak, S. (2019). The positive spillover and crossover of paternity leave use: A dyadic longitudinal analysis. *Journal of Vocational Behavior*, 115, Article 103310. https://doi.org/10 .1016/j.jvb.2019.05.007
- Kratz, F., & Brüderl, J. (2021). The age trajectory of happiness. PsyArXiv. https://doi.org/10.31234/osf.io/d8f2z
- *Ku, L. (2015). Development of materialism in adolescence: The longitudinal role of life satisfaction among Chinese youths. *Social Indicators Research*, *124*(1), 231–247. https://doi.org/10.1007/s11205-014-0787-3
- *Kuhlman, K. R., Chiang, J. J., Bower, J. E., Irwin, M. R., Cole, S. W., Dahl, R. E., Almeida, D. M., & Fuligni, A. J. (2020). Persistent low positive affect and sleep disturbance across adolescence moderate link between stress and depressive symptoms in early adulthood. *Journal of Abnormal Child Psychology*, 48(1), 109–121. https://doi.org/10.1007/s10802-019-00581-y
- Kunzmann, U., Little, T. D., & Smith, J. (2000). Is age-related stability of subjective well-being a paradox? Cross-sectional and longitudinal evidence from the Berlin Aging Study. *Psychology and Aging*, 15(3), 511–526. https://doi.org/10.1037/0882-7974.15.3.511
- Labouvie-Vief, G. (2003). Dynamic integration: Affect, cognition, and the self in adulthood. *Current Directions in Psychological Science*, 12(6), 201–206. https://doi.org/10.1046/j.0963-7214.2003.01262.x
- Labouvie-Vief, G., Diehl, M., Jain, E., & Zhang, F. (2007). Six-year change in affect optimization and affect complexity across the adult life span: A further examination. *Psychology and Aging*, 22(4), 738–751. https:// doi.org/10.1037/0882-7974.22.4.738
- *Lachman, M. E., Röcke, C., Rosnick, C., & Ryff, C. D. (2008). Realism and illusion in Americans' temporal views of their life satisfaction: Age differences in reconstructing the past and anticipating the future. *Psychological Science*, *19*(9), 889–897. https://doi.org/10.1111/j.1467-9280.2008.02173.x
- Larsen, R. J., & Diener, E. (1987). Affect intensity as an individual difference characteristic: A review. *Journal of Research in Personality*, 21(1), 1–39. https://doi.org/10.1016/0092-6566(87)90023-7

- Larson, R., Csikszentmihalyi, M., & Graef, R. (1980). Mood variability and the psychosocial adjustment of adolescents. *Journal of Youth and Adolescence*, 9(6), 469–490. https://doi.org/10.1007/BF02089885
- Lawton, M. P. (2001). Emotion in later life. *Current Directions in Psychological Science*, 10(4), 120–123. https://doi.org/10.1111/1467-8721.00130
- *Lee, S.-W., Choi, J.-S., & Lee, M. (2020). Life satisfaction and depression in the oldest old: A longitudinal study. *International Journal of Aging & Human Development*, 91(1), 37–59. https://doi.org/10.1177/0091415019843448
- *Lee, Y.-H., Cheng, C.-Y., & Lin, S. S. J. (2014). A latent profile analysis of self-control and self-esteem and the grouping effect on adolescent quality of life across two consecutive years. *Social Indicators Research*, *117*(2), 523–539. https://doi.org/10.1007/s11205-013-0360-5
- *Lemmens, J. S., Valkenburg, P. M., & Peter, J. (2011). Psychosocial causes and consequences of pathological gaming. *Computers in Human Behavior*, 27(1), 144–152. https://doi.org/10.1016/j.chb.2010.07.015
- *Lent, R. W., Miller, M. J., Smith, P. E., Watford, B. A., Hui, K., & Lim, R. H. (2015). Social cognitive model of adjustment to engineering majors: Longitudinal test across gender and race/ethnicity. *Journal of Vocational Behavior*, 86, 77–85. https://doi.org/10.1016/j.jvb.2014.11.004
- *Leung, C. Y.-W., McBride-Chang, C., & Lai, B. P.-Y. (2004). Relations among maternal parenting style, academic competence, and life satisfaction in Chinese early adolescents. *The Journal of Early Adolescence*, 24(2), 113–143. https://doi.org/10.1177/0272431603262678
- *Li, Z., Yin, X., Jiang, S., Wang, M., & Cai, T. (2014). Psychological mechanism of subjective well-being: A stable trait or situational variability. *Social Indicators Research*, 118(2), 523–534. https://doi.org/ 10.1007/s11205-013-0449-x
- *Liem, J. H., Lustig, K., & Dillon, C. (2010). Depressive symptoms and life satisfaction among emerging adults: A comparison of high school dropouts and graduates. *Journal of Adult Development*, *17*(1), 33–43. https://doi.org/10.1007/s10804-009-9076-9
- *Lima, R. F. F., & de Morais, N. A. (2019). Subjective well-being profiles of street-involved youth: A longitudinal analysis based on clusters. *Universitas Psychologica*, 18(2), 1–11. https://doi.org/10.11144/Javeriana .upsy18-2.swbp
- *Lin, W.-H., & Yi, C.-C. (2019). The effect of family cohesion and life satisfaction during adolescence on later adolescent outcomes: A prospective study. *Youth & Society*, 51(5), 680–706. https://doi.org/10 .1177/0044118X17704865
- Litwin, H., & Levinsky, M. (2022). The interplay of personality traits and social network characteristics in the subjective well-being of older adults. *Research on Aging*. Advance online publication. https://doi.org/10.1177/ 01640275221113048
- *Lopez, C. M., Felton, J. W., Driscoll, K. A., & Kistner, J. A. (2012). Brooding rumination and internalizing symptoms in childhood: Investigating symptom specificity in a multi-wave prospective study. *International Journal of Cognitive Therapy*, 5(3), 240–253. https:// doi.org/10.1521/ijct.2012.5.3.240
- Lucas, R. E., Clark, A. E., Georgellis, Y., & Diener, E. (2004). Unemployment alters the set point for life satisfaction. *Psychological Science*, 15(1), 8–13. https://doi.org/10.1111/j.0963-7214.2004.01501002.x
- Lucas, R. E., Diener, E., & Suh, E. (1996). Discriminant validity of well-being measures. *Journal of Personality and Social Psychology*, 71(3), 616–628. https://doi.org/10.1037/0022-3514.71.3.616
- Lüdtke, O., Roberts, B. W., Trautwein, U., & Nagy, G. (2011). A random walk down university avenue: Life paths, life events, and personality trait change at the transition to university life. *Journal of Personality and Social Psychology*, *101*(3), 620–637. https://doi.org/10.1037/a0023743
- Luhmann, M., Buecker, S., Kaiser, T., & Beermann, M. (2021). Nothing going on? Exploring the role of missed events in changes in subjective well-being and the Big Five personality traits. *Journal of Personality*, 89(1), 113–131. https://doi.org/10.1111/jopy.12539

- Luhmann, M., Hawkley, L. C., & Cacioppo, J. T. (2014). Thinking about one's subjective well-being: Average trends and individual differences. *Journal of Happiness Studies*, 15(4), 757–781. https://doi.org/10.1007/ s10902-013-9448-5
- Luhmann, M., Hawkley, L. C., Eid, M., & Cacioppo, J. T. (2012). Time frames and the distinction between affective and cognitive well-being. *Journal of Research in Personality*, 46(4), 431–441. https://doi.org/10 .1016/j.jrp.2012.04.004
- Luhmann, M., Hofmann, W., Eid, M., & Lucas, R. E. (2012). Subjective well-being and adaptation to life events: A meta-analysis. *Journal of Personality and Social Psychology*, 102(3), 592–615. https://doi.org/10 .1037/a0025948
- Lukkala, P. S., Honkanen, R. J., Rauma, P. H., Williams, L. J., Quirk, S. E., Kröger, H., & Koivumaa-Honkanen, H. (2016). Life satisfaction and morbidity among postmenopausal women. *PLOS ONE*, *11*(1), Article e0147521. https://doi.org/10.1371/journal.pone.0147521
- Lykken, D., & Tellegen, A. (1996). Happiness is a stochastic phenomenon. *Psychological Science*, 7(3), 186–189. https://doi.org/10.1111/j.1467-9280 .1996.tb00355.x
- *Maggiori, C., Rossier, J., Krings, F., Johnston, C., & Massoudi, K. (2016). Career pathways and professional transitions: Preliminary results from the first wave of a 7-year longitudinal study. In M. Oris, C. Roberts, D. Joye, & M. E. Stähli (Eds.), *Surveying human vulnerabilities across the life course* (pp. 131–157). Springer International Publishing. https://doi.org/10.1007/ 978-3-319-24157-9_6
- *Marques, S. C. (2016). Psychological strengths in childhood as predictors of longitudinal outcomes. *School Mental Health*, 8(3), 377–385. https:// doi.org/10.1007/s12310-016-9195-y
- *Martin, A. J., Mansour, M., Anderson, M., Gibson, R., Liem, G. A. D., & Sudmalis, D. (2013). The role of arts participation in students' academic and nonacademic outcomes: A longitudinal study of school, home, and community factors. *Journal of Educational Psychology*, 105(3), 709–727. https://doi.org/10.1037/a0032795
- *Mason, W. A., & Spoth, R. L. (2011). Longitudinal associations of alcohol involvement with subjective well-being in adolescence and prediction to alcohol problems in early adulthood. *Journal of Youth* and Adolescence, 40(9), 1215–1224. https://doi.org/10.1007/s10964-011-9632-z
- *McCabe, M. P., & Ricciardelli, L. A. (2006). A prospective study of extreme weight change behaviors among adolescent boys and girls. *Journal of Youth and Adolescence*, 35(3), 402–411. https://doi.org/10 .1007/s10964-006-9062-5
- *McCabe, M. P., & Ricciardelli, L. A. (2009). Extreme weight change behaviours: Are overweight and normal weight adolescents different, and does this vary over time? *European Eating Disorders Review*, 17(4), 301–314. https://doi.org/10.1002/erv.929
- *McLeod, G. F. H., Fergusson, D. M., John Horwood, L., & Carter, F. A. (2016). Adiposity and psychosocial outcomes at ages 30 and 35. *Social Psychiatry and Psychiatric Epidemiology*, 51(2), 309–318. https://doi.org/ 10.1007/s00127-015-1101-7
- *Metler, S. J., & Busseri, M. A. (2017). Further evaluation of the tripartite structure of subjective well-being: Evidence from longitudinal and experimental studies. *Journal of Personality*, 85(2), 192–206. https:// doi.org/10.1111/jopy.12233
- *Miller, S., Pike, J., Shono, Y., Beleva, Y., Xie, B., & Stacy, A. W. (2020). The role of negative affect in the persistence of nicotine dependence among alternative high school students: A latent growth curve analysis. *Drug and Alcohol Dependence*, 209, Article 107883. https://doi.org/10 .1016/j.drugalcdep.2020.107883
- *Mitchell, C., Reese, E., Salmon, K., & Jose, P. (2020). Narrative coherence, psychopathology, and wellbeing: Concurrent and longitudinal findings in a mid-adolescent sample. *Journal of Adolescence*, 79(1), 16–25. https:// doi.org/10.1016/j.adolescence.2019.12.003

- Moeyaert, M., Ugille, M., Natasha Beretvas, S., Ferron, J., Bunuan, R., & Van den Noortgate, W. (2017). Methods for dealing with multiple outcomes in meta-analysis: A comparison between averaging effect sizes, robust variance estimation and multilevel meta-analysis. *International Journal of Social Research Methodology*, 20(6), 559–572. https://doi.org/ 10.1080/13645579.2016.1252189
- *Molnar, D. S., Busseri, M. A., Perrier, C. P. K., & Sadava, S. W. (2009). A longitudinal examination of alcohol use and subjective well-being in an undergraduate sample. *Journal of Studies on Alcohol and Drugs*, 70(5), 704–713. https://doi.org/10.15288/jsad.2009.70.704
- Morris, S. B., & DeShon, R. P. (2002). Combining effect size estimates in meta-analysis with repeated measures and independent-groups designs. *Psychological Methods*, 7(1), 105–125. https://doi.org/10.1037/1082-989X.7.1.105
- Mroczek, D. K., & Spiro, A., III. (2005). Change in life satisfaction during adulthood: Findings from the veterans affairs normative aging study. *Journal of Personality and Social Psychology*, 88(1), 189–202. https:// doi.org/10.1037/0022-3514.88.1.189
- *Müller, D., Ziegelmann, J. P., Simonson, J., Tesch-Römer, C., & Huxhold, O. (2014). Volunteering and subjective well-being in later adulthood: Is self-efficacy the key? *International Journal of Developmental Science*, 8(3–4), 125–135. https://doi.org/10.3233/DEV-14140
- *Mulvihill, K. (2021). A self-determination theory perspective on wellbeing in the transition from university: Trait autonomy and basic psychological need satisfaction through graduates' goal pursuits, past and present. Concordia University Montréal.
- Mund, M., Freuding, M. M., Möbius, K., Horn, N., & Neyer, F. J. (2020). The stability and change of loneliness across the life span: A meta-analysis of longitudinal studies. *Personality and Social Psychology Review*, 24(1), 24–52. https://doi.org/10.1177/1088868319850738
- *Nelis, S., Bastin, M., Raes, F., Mezulis, A., & Bijttebier, P. (2016). Trait affectivity and response styles to positive affect: Negative affectivity relates to dampening and positive affectivity relates to enhancing. *Personality and Individual Differences*, 96, 148–154. https://doi.org/10 .1016/j.paid.2016.02.087
- Nes, R. B., & Røysamb, E. (2015). The heritability of subjective well-being: Review and meta-analysis. In M. Pluess (Ed.), *Genetics of psychological well-being: The role of heritability and genetics* (pp. 75–96). Oxford University Press. https://doi.org/10.1093/acprof:oso/9780199686674.003.0005
- *Ng, Z. J., Huebner, E. S., Hills, K. J., & Valois, R. F. (2018). Mediating effects of emotion regulation strategies in the relations between stressful life events and life satisfaction in a longitudinal sample of early adolescents. *Journal of School Psychology*, 70, 16–26. https://doi.org/10 .1016/j.jsp.2018.06.001
- *Opree, S. J., Buijzen, M., & Valkenburg, P. M. (2012). Lower life satisfaction related to materialism in children frequently exposed to advertising. *Pediatrics*, 130(3), e486–e491. https://doi.org/10.1542/peds .2011-3148
- *Orth, U., Erol, R. Y., Ledermann, T., & Grob, A. (2018). Codevelopment of well-being and self-esteem in romantic partners: Disentangling the effects of mutual influence and shared environment. *Developmental Psychology*, 54(1), 151–166. https://doi.org/10.1037/dev0000400
- Orth, U., Erol, R. Y., & Luciano, E. C. (2018). Development of self-esteem from age 4 to 94 years: A meta-analysis of longitudinal studies. *Psychological Bulletin*, 144(10), 1045–1080. https://doi.org/10.1037/bu 10000161
- *Orth, U., & Luciano, E. C. (2015). Self-esteem, narcissism, and stressful life events: Testing for selection and socialization. *Journal of Personality* and Social Psychology, 109(4), 707–721. https://doi.org/10.1037/pspp 0000049
- *Orth, U., Robins, R. W., & Widaman, K. F. (2012). Life-span development of self-esteem and its effects on important life outcomes. *Journal of Personality and Social Psychology*, *102*(6), 1271–1288. https://doi.org/10 .1037/a0025558

- *Orth, U., Trzesniewski, K. H., & Robins, R. W. (2010). Self-esteem development from young adulthood to old age: A cohort-sequential longitudinal study. *Journal of Personality and Social Psychology*, 98(4), 645–658. https://doi.org/10.1037/a0018769
- *Otterbach, S., Sousa-Poza, A., & Møller, V. (2018). A cohort analysis of subjective wellbeing and ageing: Heading towards a midlife crisis? *Longitudinal and Life Course Studies*, 9(4), 382–411. https://doi.org/10 .14301/Ilcs.v9i4.509
- Owen, R., Berry, K., & Brown, L. J. E. (2021). Enhancing older adults' wellbeing and quality of life through purposeful activity: A systematic review of intervention studies. *The Gerontologist*, 62(6), e317–e327. https:// doi.org/10.1093/geront/gnab017
- Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D., Shamseer, L., Tetzlaff, J. M., Akl, E. A., Brennan, S. E., Chou, R., Glanville, J., Grimshaw, J. M., Hróbjartsson, A., Lalu, M. M., Li, T., Loder, E. W., Mayo-Wilson, E., McDonald, S., ... Moher, D. (2021). The PRISMA 2020 statement: An updated guideline for reporting systematic reviews. *PLOS Medicine*, *18*(3), Article e1003583. https:// doi.org/10.1371/journal.pmed.1003583
- *Parra, A., Oliva, A., & Sánchez-Queija, I. (2015). Development of emotional autonomy from adolescence to young adulthood in Spain. *Journal of Adolescence*, 38(1), 57–67. https://doi.org/10.1016/j.adolescence.2014.11.003
- *Peluso, M. A. M., Savalli, C., Cúri, M., Gorenstein, C., & Andrade, L. H. (2010). Mood changes in the course of preparation for the Brazilian university admission exam—A longitudinal study. *Revista Brasileira de Psiquiatria*, 32(1), 30–36. https://doi.org/10.1590/S1516-44462010000100007
- Pinquart, M. (2001). Age differences in perceived positive affect, negative affect, and affect balance in middle and old age. *Journal of Happiness Studies*, 2(4), 375–405. https://doi.org/10.1023/A:1013938001116
- Piper, A. T. (2015). Sliding down the U-shape? A dynamic panel investigation of the age-well-being relationship, focusing on young adults. *Social Science & Medicine*, 143, 54–61. https://doi.org/10.1016/j .socscimed.2015.08.042
- *Planalp, E. M., Van Hulle, C., Lemery-Chalfant, K., & Goldsmith, H. H. (2017). Genetic and environmental contributions to the development of positive affect in infancy. *Emotion*, 17(3), 412–420. https://doi.org/10 .1037/emo0000238
- *Poulin, M. J., Silver, R. C., Gil-Rivas, V., Holman, E. A., & McIntosh, D. N. (2009). Finding social benefits after a collective trauma: Perceiving societal changes and well-being following 9/11. *Journal of Traumatic Stress*, 22(2), 81–90. https://doi.org/10.1002/jts.20391
- *Powdthavee, N., & Vignoles, A. (2008). Mental health of parents and life satisfaction of children: A within-family analysis of intergenerational transmission of well-being. *Social Indicators Research*, 88(3), 397–422. https://doi.org/10.1007/s11205-007-9223-2
- Pusch, S., Mund, M., Hagemeyer, B., & Finn, C. (2019). Personality development in emerging and young adulthood: A study of age differences. *European Journal of Personality*, 33(3), 245–263. https:// doi.org/10.1002/per.2181
- *Pushkar, D., Bye, D., Conway, M., Wrosch, C., Chaikelson, J., Etezadi, J., Giannopoulos, C., Li, K., & Tabri, N. (2014). Does child gender predict older parents' well-being? *Social Indicators Research*, *118*(1), 285–303. https://doi.org/10.1007/s11205-013-0403-y
- *Räikkönen, E., Kokko, K., & Rantanen, J. (2011). Timing of adult transitions: Antecedents and implications for psychological functioning. *European Psychologist*, 16(4), 314–323. https://doi.org/10.1027/1016-9040/a000050
- *Rangul, V., Bauman, A., Holmen, T. L., & Midthjell, K. (2012). Is physical activity maintenance from adolescence to young adulthood associated with reduced CVD risk factors, improved mental health and satisfaction with life: The HUNT Study, Norway. *The International Journal of Behavioral Nutrition and Physical Activity*, 9(1), Article 144. https:// doi.org/10.1186/1479-5868-9-144

- *Ranta, M., Chow, A., & Salmela-Aro, K. (2013). Trajectories of life satisfaction and the financial situation in the transition to adulthood. *Longitudinal and Life Course Studies*, 4(1), 57–77. https://doi.org/10.14301/ llcs.v4i1.216
- *Ranta, M., & Salmela-Aro, K. (2018). Subjective financial situation and financial capability of young adults in Finland. *International Journal* of Behavioral Development, 42(6), 525–534. https://doi.org/10.1177/ 0165025417745382
- Raudenbush, S. W. (2009). Analyzing effect sizes: Random-effects models. In H. Cooper, L. V. Hedges, & J. C. Valentine (Eds.), *The handbook of research synthesis and meta-analysis* (2nd ed., pp. 295–316). Sage Publications.
- *Rauma, P. H., Koivumaa-Honkanen, H., Williams, L. J., Tuppurainen, M. T., Kröger, H. P., & Honkanen, R. J. (2014). Life satisfaction and bone mineral density among postmenopausal women: Cross-sectional and longitudinal associations. *Psychosomatic Medicine*, 76(9), 709–715. https://doi.org/10.1097/PSY.000000000000114
- R Core Team. (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing. https://www.r-project.org/
- *Recksiedler, C., Settersten, R. A., Jr., Geldhof, G. J., & Hooker, K. (2019). Stable goals despite economic strain: Young adults' goal appraisals across the great recession. *International Journal of Behavioral Development*, 43(2), 147–156. https://doi.org/10.1177/0165025418798494
- *Reis, D., Friese, M., Hart, A., & Krautter, K. (2021). Projekt: Alles anders? [Unpublished raw data]. Saarland University.
- *Reiter, S. F., Bjørk, M. H., Daltveit, A. K., Veiby, G., Kolstad, E., Engelsen, B. A., & Gilhus, N. E. (2016). Life satisfaction in women with epilepsy during and after pregnancy. *Epilepsy & Behavior*, 62, 251–257. https:// doi.org/10.1016/j.yebeh.2016.06.025
- *Ricciardelli, L. A., McCabe, M. P., Lillis, J., & Thomas, K. (2006). A longitudinal investigation of the development of weight and muscle concerns among preadolescent boys. *Journal of Youth and Adolescence*, 35(2), 177–187. https://doi.org/10.1007/s10964-005-9004-7
- *Richmond, M. J., Mermelstein, R. J., & Wakschlag, L. S. (2013). Direct observations of parenting and real-time negative affect among adolescent smokers and nonsmokers. *Journal of Clinical Child and Adolescent Psychology*, 42(5), 617–628. https://doi.org/10.1080/15374416.2012 .738452
- *Richter, E. P., Brähler, E., Stöbel-Richter, Y., Zenger, M., & Berth, H. (2020). The long-lasting impact of unemployment on life satisfaction: Results of a longitudinal study over 20 years in East Germany. *Health* and Quality of Life Outcomes, 18(1), Article 361. https://doi.org/10.1186/ s12955-020-01608-5
- Roberts, B. W., & Robins, R. W. (2021). Personality development across the life course: A neo-socioanalytic perspective. In O. P. John & R. W. Robins (Eds.), *Handbook of personality: Theory and research* (pp. 259–283). Guilford Press.
- Roberts, B. W., Walton, K. E., & Viechtbauer, W. (2006). Patterns of meanlevel change in personality traits across the life course: A meta-analysis of longitudinal studies. *Psychological Bulletin*, 132(1), 1–25. https://doi.org/ 10.1037/0033-2909.132.1.1
- Roberts, B. W., & Wood, D. (2006). Personality development in the context of the neo-socioanalytic model of personality. In D. K. Mroczek & T. D. Little (Eds.), *Handbook of personality development* (pp. 11–39). Lawrence Erlbaum.
- Robinson, M. D., & Clore, G. L. (2002). Belief and feeling: Evidence for an accessibility model of emotional self-report. *Psychological Bulletin*, *128*(6), 934–960. https://doi.org/10.1037/0033-2909.128.6.934
- Sainz, M., Martínez, R., Moya, M., Rodríguez-Bailón, R., & Vaes, J. (2021). Lacking socio-economic status reduces subjective well-being through perceptions of meta-dehumanization. *British Journal of Social Psychology*, 60(2), 470–489. https://doi.org/10.1111/bjso.12412

- *Salmela-Aro, K., & Tynkkynen, L. (2010). Trajectories of life satisfaction across the transition to post-compulsory education: Do adolescents follow different pathways? *Journal of Youth and Adolescence*, 39(8), 870–881. https://doi.org/10.1007/s10964-009-9464-2
- *Sánchez-Álvarez, N., Extremera, N., & Fernández-Berrocal, P. (2015). Maintaining life satisfaction in adolescence: Affective mediators of the influence of perceived emotional intelligence on overall life satisfaction judgments in a two-year longitudinal study. *Frontiers in Psychology*, 6, Article 1892. https://doi.org/10.3389/fpsyg.2015.01892
- *Scharkow, M., Festl, R., & Quandt, T. (2014). Longitudinal patterns of problematic computer game use among adolescents and adults—A 2-year panel study. *Addiction*, 109(11), 1910–1917. https://doi.org/10.1111/add .12662
- Schimmack, U., Schupp, J., & Wagner, G. G. (2008). The influence of environment and personality on the affective and cognitive component of subjective well-being. *Social Indicators Research*, 89(1), 41–60. https:// doi.org/10.1007/s11205-007-9230-3
- *Schmiedeberg, C., Huyer-May, B., Castiglioni, L., & Johnson, M. D. (2017). The more or the better? How sex contributes to life satisfaction. *Archives of Sexual Behavior*, 46(2), 465–473. https://doi.org/10.1007/ s10508-016-0843-y
- *Seery, M. D., Holman, E. A., & Silver, R. C. (2010). Whatever does not kill us: Cumulative lifetime adversity, vulnerability, and resilience. *Journal of Personality and Social Psychology*, 99(6), 1025–1041. https://doi.org/10 .1037/a0021344
- Shankar, A., Rafnsson, S. B., & Steptoe, A. (2015). Longitudinal associations between social connections and subjective wellbeing in the English Longitudinal Study of Ageing. *Psychology & Health*, 30(6), 686–698. https://doi.org/10.1080/08870446.2014.979823
- *Shek, D. T. L. (2007). A longitudinal study of perceived parental psychological control and psychological well-being in Chinese adolescents in Hong Kong. *Journal of Clinical Psychology*, 63(1), 1–22. https:// doi.org/10.1002/jclp.20331
- *Shek, D. T. L., & Li, X. (2016). Perceived school performance, life satisfaction, and hopelessness: A 4-year longitudinal study of adolescents in Hong Kong. *Social Indicators Research*, *126*(2), 921–934. https:// doi.org/10.1007/s11205-015-0904-y
- *Shek, D. T. L., & Liu, T. T. (2014). Life satisfaction in junior secondary school students in Hong Kong: A 3-year longitudinal study. *Social Indicators Research*, 117(3), 777–794. https://doi.org/10.1007/s11205-013-0398-4
- *Shoshani, A., & Slone, M. (2013). Middle school transition from the strengths perspective: Young adolescents' character strengths, subjective well-being, and school adjustment. *Journal of Happiness Studies*, 14(4), 1163–1181. https://doi.org/10.1007/s10902-012-9374-y
- *Shoshani, A., Steinmetz, S., & Kanat-Maymon, Y. (2016). Effects of the Maytiv positive psychology school program on early adolescents' well-being, engagement, and achievement. *Journal of School Psychology*, 57, 73–92. https://doi.org/10.1016/j.jsp.2016.05.003
- *Song, J., Ip, K. I., Yan, J., Lui, P. P., Kamata, A., & Kim, S. Y. (2022). Pathways linking ethnic discrimination and drug-using peer affiliation to underage drinking status among Mexican-origin adolescents. *Experimental* and Clinical Psychopharmacology, 30(5), 609–619. https://doi.org/10 .1037/pha0000504
- Soto, C. J. (2015). Is happiness good for your personality? Concurrent and prospective relations of the big five with subjective well-being. *Journal of Personality*, 83(1), 45–55. https://doi.org/10.1111/jopy.12081
- Soto, C. J. (2016). The little six personality dimensions from early childhood to early adulthood: Mean-level age and gender differences in parents' reports. *Journal of Personality*, 84(4), 409–422. https://doi.org/10.1111/jo py.12168
- Soto, C. J., John, O. P., Gosling, S. D., & Potter, J. (2011). Age differences in personality traits from 10 to 65: Big Five domains and facets in a large

cross-sectional sample. *Journal of Personality and Social Psychology*, 100(2), 330–348. https://doi.org/10.1037/a0021717

- Soto, C. J., & Tackett, J. L. (2015). Personality traits in childhood and adolescence: Structure, development, and outcomes. *Current Directions* in *Psychological Science*, 24(5), 358–362. https://doi.org/10.1177/09637 21415589345
- *Sowden, W. J. (2021). Culture, performance, and health study (Walter Reed Army Institute of Research Protocol 2460) [Unpublished raw data]. Uniformed Services University.
- Specht, J., Egloff, B., & Schmukle, S. C. (2011). Stability and change of personality across the life course: The impact of age and major life events on mean-level and rank-order stability of the Big Five. *Journal of Personality and Social Psychology*, 101(4), 862–882. https://doi.org/10 .1037/a0024950
- *Spiegler, O., Wölfer, R., & Hewstone, M. (2019). Dual identity development and adjustment in Muslim minority adolescents. *Journal of Youth and Adolescence*, 48(10), 1924–1937. https://doi.org/10.1007/s10964-019-01117-9
- *Spindler, G., Stopsack, M., Aldinger, M., Grabe, H. J., & Barnow, S. (2016). What about the "ups and downs" in our daily life? The influence of affective instability on mental health. *Motivation and Emotion*, 40(1), 148–161. https://doi.org/10.1007/s11031-015-9509-7
- *Spuling, S. M., Wurm, S., Tesch-Römer, C., & Huxhold, O. (2015). Changing predictors of self-rated health: Disentangling age and cohort effects. *Psychology and Aging*, 30(2), 462–474. https://doi.org/10.1037/a0039111
- *Steger, M. F., & Kashdan, T. B. (2007). Stability and specificity of meaning in life and life satisfaction over one year. *Journal of Happiness Studies*, 8(2), 161–179. https://doi.org/10.1007/s10902-006-9011-8
- Steinmayr, R., Wirthwein, L., Modler, L., & Barry, M. M. (2019). Development of subjective well-being in adolescence. *International Journal of Environmental Research and Public Health*, 16(19), Article 3690. https://doi.org/10.3390/ijerph16193690
- Sterne, J. A. C., & Egger, M. (2005). Regression methods to detect publication and other bias in meta-analysis. In H. R. Rothstein, J. A. Sutton, & M. Borenstein (Eds.), *Publication bias in meta-analysis— Prevention, assessment and adjustments* (pp. 99–110). Wiley. https:// doi.org/10.1002/0470870168.ch6
- *Stice, E., Shaw, H., Burton, E., & Wade, E. (2006). Dissonance and healthy weight eating disorder prevention programs: A randomized efficacy trial. *Journal of Consulting and Clinical Psychology*, 74(2), 263–275. https:// doi.org/10.1037/0022-006X.74.2.263
- Stone, A. A., Schwartz, J. E., Broderick, J. E., & Deaton, A. (2010). A snapshot of the age distribution of psychological well-being in the United States. *Proceedings of the National Academy of Sciences of the United States of America*, 107(22), 9985–9990. https://doi.org/10.1073/pnas .1003744107
- *Strumpp, T. (2006). Anpassung der Bindung an persönliche berufliche Ziele als Erfolgsfaktor organisationaler Sozialisation: Latente Wachstumskurvenanalysen bei Berufseinsteigern. Universität Bielefeld.
- *Stutts, L. A., Leary, M. R., Zeveney, A. S., & Hufnagle, A. S. (2018). A longitudinal analysis of the relationship between self-compassion and the psychological effects of perceived stress. *Self and Identity*, *17*(6), 609–626. https://doi.org/10.1080/15298868.2017.1422537
- *Su, S., Li, X., Lin, D., & Zhu, M. (2017). Future orientation, social support, and psychological adjustment among left-behind children in rural China: A longitudinal study. *Frontiers in Psychology*, 8, Article 1309. https:// doi.org/10.3389/fpsyg.2017.01309
- *Suldo, S. M., & Huebner, E. S. (2004). Does life satisfaction moderate the effects of stressful life events on psychopathological behavior during adolescence? *School Psychology Quarterly*, *19*(2), 93–105. https:// doi.org/10.1521/scpq.19.2.93.33313
- *Sun, R. C. F., & Shek, D. T. L. (2013). Longitudinal influences of positive youth development and life satisfaction on problem behaviour among

adolescents in Hong Kong. Social Indicators Research, 114(3), 1171-1197. https://doi.org/10.1007/s11205-012-0196-4

- Tanner-Smith, E. E., & Tipton, E. (2014). Robust variance estimation with dependent effect sizes: Practical considerations including a software tutorial in Stata and spss. *Research Synthesis Methods*, 5(1), 13–30. https://doi.org/10.1002/jrsm.1091
- *Taradash, A. R. (2006). Romantic relationships in adolescence: Influences on negative affect and minor misbehavior. York University.
- *Tauber, B., Wahl, H.-W., & Schröder, J. (2016). Personality and life satisfaction over 12 years: Contrasting mid- and late life. *GeroPsych*, 29(1), 37–48. https://doi.org/10.1024/1662-9647/a000141
- *Taylor, A. M., Goldberg, C., Shore, L. M., & Lipka, P. (2008). The effects of retirement expectations and social support on post-retirement adjustment: A longitudinal analysis. *Journal of Managerial Psychology*, 23(4), 458–470. https://doi.org/10.1108/02683940810869051
- *Teng, Z., Pontes, H. M., Nie, Q., Xiang, G., Griffiths, M. D., & Guo, C. (2020). Internet gaming disorder and psychosocial well-being: A longitudinal study of older-aged adolescents and emerging adults. *Addictive Behaviors*, 110, Article 106530. https://doi.org/10.1016/j.addbeh .2020.106530
- *Thomsen, D. K., Mehlsen, M. Y., Olesen, F., Hokland, M., Viidik, A., Avlund, K., & Zachariae, R. (2004). Is there an association between rumination and self-reported physical health? A one-year follow-up in a young and an elderly sample. *Journal of Behavioral Medicine*, 27(3), 215–231. https://doi.org/10.1023/B:JOBM.0000028496.41492.34
- Tov, W. (2018). Well-being concepts and components. In E. Diener, S. Oishi, & L. Tay (Eds.), *Handbook of well-being* (pp. 1–15). DEF Publishers.
- *Trepte, S., & Dienlin, T. (2020). Privacy Longitudinal Study [Unpublished raw data]. University of Hohenheim. https://doi.org/10.7802/2117
- *Tumminia, M. J., Colaianne, B. A., Roeser, R. W., & Galla, B. M. (2020). How is mindfulness linked to negative and positive affect? Rumination as an explanatory process in a prospective longitudinal study of adolescents. *Journal of Youth and Adolescence*, 49(10), 2136–2148. https://doi.org/10 .1007/s10964-020-01238-6
- Urry, H. L., & Gross, J. J. (2010). Emotion regulation in older age. Current Directions in Psychological Science, 19(6), 352–357. https://doi.org/10 .1177/0963721410388395
- *Valle, M. F., Huebner, E. S., & Suldo, S. M. (2006). An analysis of hope as a psychological strength. *Journal of School Psychology*, 44(5), 393–406. https://doi.org/10.1016/j.jsp.2006.03.005
- Van den Akker, A. L., Deković, M., Asscher, J., & Prinzie, P. (2014). Meanlevel personality development across childhood and adolescence: A temporary defiance of the maturity principle and bidirectional associations with parenting. *Journal of Personality and Social Psychology*, 107(4), 736–750. https://doi.org/10.1037/a0037248
- *van den Eijnden, R., Koning, I., Doornwaard, S., van Gurp, F., & Ter Bogt, T. (2018). The impact of heavy and disordered use of games and social media on adolescents' psychological, social, and school functioning. *Journal of Behavioral Addictions*, 7(3), 697–706. https://doi.org/10.1556/ 2006.7.2018.65
- Viechtbauer, W. (2010). Conducting meta-analyses in *R* with the metafor package. *Journal of Statistical Software*, 36(3), 1–48. https://doi.org/10 .18637/jss.v036.i03
- Viechtbauer, W. (2019). An introduction to longitudinal meta-analysis in R with the metafor package [Conference session]. Lifespan Social-Personality Preconference, Society for Personality and Social Psychology (SPSP) Annual Convention, Portland, OR, United States. https://www .wvbauer.com/doku.php/presentations
- Viechtbauer, W., & Cheung, M. W.-L. (2010). Outlier and influence diagnostics for meta-analysis. *Research Synthesis Methods*, 1(2), 112– 125. https://doi.org/10.1002/jrsm.11
- *Viegas, L. M., Bermeitinger, C., & Greve, W. (2021). aera: Adaptation and Evolution of regulatory processes in old [Unpublished raw data]. The Age.

- Vogel, N., Schilling, O. K., Wahl, H.-W., Beekman, A. T. F., & Penninx, B. W. J. H. (2013). Time-to-death-related change in positive and negative affect among older adults approaching the end of life. *Psychology and Aging*, 28(1), 128–141. https://doi.org/10.1037/a0030471
- *Vrangalova, Z. (2015). Hooking up and psychological well-being in college students: Short-term prospective links across different hookup definitions. *Journal of Sex Research*, 52(5), 485–498. https://doi.org/10.1080/ 00224499.2014.910745
- *Waggel, S. E., Lipnicki, D. M., Delbaere, K., Kochan, N. A., Draper, B., Andrews, G., Sachdev, P. S., & Brodaty, H. (2015). Neuroticism scores increase with late-life cognitive decline. *International Journal of Geriatric Psychiatry*, 30(9), 985–993. https://doi.org/10.1002/gps.4251
- *Wagner, J., Becker, M., Lüdtke, O., & Trautwein, U. (2015). The first partnership experience and personality development: A propensity score matching study in young adulthood. *Social Psychological & Personality Science*, 6(4), 455–463. https://doi.org/10.1177/1948550614566092
- Wahlstrom, D., Collins, P., White, T., & Luciana, M. (2010). Developmental changes in dopamine neurotransmission in adolescence: Behavioral implications and issues in assessment. *Brain and Cognition*, 72(1), 146–159. https://doi.org/10.1016/j.bandc.2009.10.013
- *Wakefield, J. R. H., Sani, F., & Herrera, M. (2018). Greater university identification-but not greater contact-leads to more life satisfaction: Evidence from a Spanish longitudinal study. *Applied Psychology. Health* and Well-Being, 10(2), 330–344. https://doi.org/10.1111/aphw.12125
- *Walters, G. D., & Espelage, D. L. (2018). Exploring the victimization-early substance misuse relationship: In search of moderating and mediating effects. *Child Abuse & Neglect*, 81, 354–365. https://doi.org/10.1016/j .chiabu.2018.05.006
- *Wang, L., Crawford, J. D., Reppermund, S., Trollor, J., Campbell, L., Baune, B. T., Sachdev, P., Brodaty, H., Samaras, K., & Smith, E. (2018). Body mass index and waist circumference predict health-related quality of life, but not satisfaction with life, in the elderly. *Quality of Life Research*, 27(10), 2653–2665. https://doi.org/10.1007/s11136-018-1904-6
- *Wang, R., Liu, H., Jiang, J., & Song, Y. (2017). Will materialism lead to happiness? A longitudinal analysis of the mediating role of psychological needs satisfaction. *Personality and Individual Differences*, 105, 312–317. https://doi.org/10.1016/j.paid.2016.10.014
- Watson, D., Clark, L. A., & Tellegen, A. (1988). Development and validation of brief measures of positive and negative affect: The PANAS scales. *Journal of Personality and Social Psychology*, 54(6), 1063–1070. https:// doi.org/10.1037/0022-3514.54.6.1063
- *Weinstein, S. M., Mermelstein, R. J., Hedeker, D., Hankin, B. L., & Flay, B. R. (2006). The time-varying influences of peer and family support on adolescent daily positive and negative affect. *Journal of Clinical Child* and Adolescent Psychology, 35(3), 420–430. https://doi.org/10.1207/ s15374424jccp3503_7
- *Wickrama, K. A. S., Lee, S., Klopack, E. T., & Wickrama, T. (2019). Stressful work conditions, positive affect, and physical health of middleaged couples: A dyadic analysis. *Stress and Health*, 35(4), 382–395. https://doi.org/10.1002/smi.2866
- *Wiese, B., & Freund, A. (2005). Goal progress makes one happy, or does it? Longitudinal findings from the work domain. *Journal of Occupational* and Organizational Psychology, 78(2), 287–304. https://doi.org/10.1348/ 096317905X26714
- *Williams, K. E., Ciarrochi, J., & Heaven, P. C. L. (2015). Relationships between valued action and well-being across the transition from high school to early adulthood. *The Journal of Positive Psychology*, *10*(2), 127–140. https://doi.org/10.1080/17439760.2014.920404
- *Willroth, E. C., Atherton, O. E., & Robins, R. W. (2021). Life satisfaction trajectories during adolescence and the transition to young adulthood: Findings from a longitudinal study of Mexican-origin youth. *Journal of Personality and Social Psychology*, *120*(1), 192–205. https://doi.org/10 .1037/pspp0000294

- Wilson, R. S., Wang, T., Yu, L., Bennett, D. A., & Boyle, P. A. (2020). Normative cognitive decline in old age. *Annals of Neurology*, 87(6), 816–829. https://doi.org/10.1002/ana.25711
- Wrzus, C., Hänel, M., Wagner, J., & Neyer, F. J. (2013). Social network changes and life events across the life span: A meta-analysis. *Psychological Bulletin*, 139(1), 53–80. https://doi.org/10.1037/a0028601
- Wrzus, C., Müller, V., Wagner, G. G., Lindenberger, U., & Riediger, M. (2013). Affective and cardiovascular responding to unpleasant events from adolescence to old age: Complexity of events matters. *Developmental Psychology*, 49(2), 384–397. https://doi.org/10.1037/a0028325
- *Wu, L.-H., & Tsay, R.-M. (2018). The search for happiness: Work experiences and quality of life of older Taiwanese men. *Social Indicators Research*, 136(3), 1031–1051. https://doi.org/10.1007/s11205-016-1531-y
- *Wu, X., Gai, X., & Wang, W. (2020). Subjective well-being and academic performance among middle schoolers: A two-wave longitudinal study. *Journal of Adolescence*, 84(1), 11–22. https://doi.org/10.1016/j.adolesce nce.2020.07.011
- Wunder, C., Wiencierz, A., Schwarze, J., & Küchenhoff, H. (2013). Well-Being over the life span: Semiparametric evidence from British and German Longitudinal Data. *The Review of Economics and Statistics*, 95(1), 154–167. https://doi.org/10.1162/REST_a_00222
- Wurm, S., Diehl, M., Kornadt, A. E., Westerhof, G. J., & Wahl, H.-W. (2017). How do views on aging affect health outcomes in adulthood and late life? Explanations for an established connection. *Developmental Review*, 46, 27–43. https://doi.org/10.1016/j.dr.2017.08.002
- *Yeatman, S., & Smith-Greenaway, E. (2018). Birth planning and women's and men's health in Malawi. *Studies in Family Planning*, 49(3), 213–235. https://doi.org/10.1111/sifp.12060
- *Youssef, A.-J. S. (2014). Gratitude's relation to well-being: A four-year longitudinal investigation in late adolescents. Hofstra University.
- *Zank, S., Woopen, C., Wagner, M., Rietz, C., & Kaspar, R. (2022). Lebensqualität und Wohlbefinden hochaltriger Menschen in NRW (Repräsentativbefragung NRW80+) [Raw data: GESIS Datenarchiv, Köln. ZA7558 Datenfile Version 1.0.0]. https://doi.org/10.4232/1.13527
- Zannella, M., Hammer, B., Prskawetz, A., & Sambt, J. (2019). A quantitative assessment of the rush hour of life in Austria, Italy and Slovenia. European *Journal of Population*, 35(4), 751–776. https://doi.org/10.1007/s10680-018-9502-4
- *Zhan, Y., Froidevaux, A., Li, Y., Wang, M., & Shi, J. (2022). Preretirement resources and postretirement life satisfaction change trajectory: Examining the mediating role of retiree experience during retirement transition phase. *Journal of Applied Psychology*. Advance online publication. https://doi.org/10.1037/apl0001043
- *Zhang, G., Eisenberg, N., Liang, Z., Li, Y., & Deng, H. (2017). The relations of migrant status and parenting to Chinese adolescents' adjustment. *International Journal of Behavioral Development*, 41(3), 320–328. https://doi.org/10.1177/0165025415627699
- *Zhang, W., Braun, K. L., & Wu, Y. Y. (2017). The educational, racial and gender crossovers in life satisfaction: Findings from the longitudinal Health and Retirement Study. *Archives of Gerontology and Geriatrics*, 73, 60–68. https://doi.org/10.1016/j.archger.2017.07.014
- *Zhu, X., & Shek, D. T. L. (2020). Predictive effect of positive youth development attributes on delinquency among adolescents in mainland China. *Frontiers in Psychology*, *11*, Article 615900. https://doi.org/10 .3389/fpsyg.2020.615900
- *Zhu, X., & Shek, D. T. L. (2021). Parental factors and adolescent well-being: Associations between developmental trajectories. *Children and Youth Services Review*, 127, Article 106071. https://doi.org/10.1016/j.childyouth.2021.106071
- Zimmermann, P., & Iwanski, A. (2014). Emotion regulation from early adolescence to emerging adulthood and middle adulthood: Age differences, gender differences, and emotion-specific developmental variations. *International Journal of Behavioral Development*, 38(2), 182–194. https://doi.org/10.1177/0165025413515405

446

- *Zink, J., Ebrahimian, S., Belcher, B. R., & Leventhal, A. M. (2020). Reciprocal associations between depression and screen-based sedentary behaviors in adolescents differ by depressive symptom dimension and screen-type. *Journal of Affective Disorders*, 263, 39–46. https://doi.org/ 10.1016/j.jad.2019.11.130
- *Zygar-Hoffmann, C., Cludius, B., Werner, G. G., Stefan, A., Sckopke, P., & Schönbrodt, F. D. (2023). Unpublished raw data on the Corona couple ESM study [Unpublished raw data]. LMU Munich. https://doi.org/10.17605/OSF .IO/72EJG
- *Zygar-Hoffmann, C., Hagemeyer, B., Pusch, S., & Schönbrodt, F. D. (2020). A large longitudinal study on motivation, behavior and

satisfaction in couples: Research data from a four-week experience sampling study with a pre-, post-, and one-year follow-up assessment [Unpublished raw data]. LMU Munich. https://doi.org/10.5160/psychdata .zrce18mo99_v20000

> Received April 12, 2022 Revision received June 8, 2023 Accepted June 19, 2023

E-Mail Notification of Your Latest Issue Online!

Would you like to know when the next issue of your favorite APA journal will be available online? This service is now available to you. Sign up at https://my.apa.org/portal/alerts/ and you will be notified by e-mail when issues of interest to you become available!